# UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

## Form S-8

REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933

## AMERICAN INTERNATIONAL GROUP, INC.

(Exact name of registrant as specified in its charter)

**Delaware** 

(State or other jurisdiction of incorporation)

13-2592361 (IRS Employer identification No.)

70 Pine Street, New York, New York 10270

(Address, including zip code, of principal executive offices)

American International Group, Inc. Amended and Restated 2007 Stock Incentive Plan (Full title of the plan)

Kathleen E. Shannon
Senior Vice President, Secretary and Deputy General Counsel
70 Pine Street
New York, New York 10270
(212) 770-7000

(Name, address, including zip code, and telephone number, including area code, of agent for service)

Title of Securities to be Registered	Amount to be Registered (1)	Proposed Maximum Offering Price Per Share (2)	Proposed Maximum Aggregate Offering Price (2)	Amount of Registration Fee (1) (2)
Common Stock, par value \$2.50 per share	135,000,000 shares	\$55.84	\$7,538,400,000	\$231,429.00

- 1. This Registration Statement also relates to an indeterminate number of additional shares of Common Stock that may be issued pursuant to anti-dilution and adjustment provisions of the above-referenced plan.
- Estimated solely for purposes of calculating the registration fee. This estimate has been computed in accordance with Rule 457(c) and (h)(1) and is calculated based upon the average of the high and low sales prices of the Common Stock of American International Group, Inc. on December 17, 2007, as reported on the New York Stock Exchange Composite Tape.

## **TABLE OF CONTENTS**

<b>PART</b>	Ι
DART	Ħ

Item 3. Incorporation of Documents by Reference

Item 4. Description of Securities

Item 5. Interests of Named Experts and Counsel

Item 6. Indemnification of Directors and Officers

Item 7. Exemption From Registration Claimed

Item 8. Exhibits

Item 9. Undertakings

**SIGNATURES** 

**EXHIBIT INDEX** 

EX-4: AMENDED AND RESTATED 2007 STOCK INCENTIVE PLAN

EX-5: OPINION OF KATHLEEN E. SHANNON RE VALIDITY

EX-10.A: FORM OF STOCK OPTION AWARD AGREEMENT

EX-10.B: FORM OF PERFORMANCE RSU AWARD AGREEMENT

EX-10.C: FORM OF TIME-VESTED RSU AWARD AGREEMENT

EX-10.D: FORM OF TIME VESTED RSU AWARD AGREEMENT WITH FOUR-YEAR PRO RATA VESTING

EX-10.E: FORM OF TIME-VESTED RSU AWARD AGREEMENT WITH THREE-YEAR PRO RATA VESTING

EX-10.F: FORM OF NON-EMPLOYEE DIRECTOR DEFERRED STOCK UNITS AWARD AGREEMENT

EX-23.A: CONSENT OF PRICEWATERHOUSECOOPERS LLP

## PART I

## INFORMATION REQUIRED IN THE SECTION 10(a) PROSPECTUS

All information required by Part I to be contained in the prospectus is omitted from this Registration Statement in accordance with Rule 428 under the Securities Act of 1933, as amended (the "Securities Act").

#### AMERICAN INTERNATIONAL GROUP, INC.

70 Pine Street New York, New York 10270 (212) 770-7000 Common Stock, \$2.50 Par Value

This prospectus relates to up to 135,000,000 common shares which may be delivered pursuant to the terms of awards under the American International Group, Inc. Amended and Restated 2007 Stock Incentive Plan (the "Plan"). The issuer of the shares of common stock, par value \$2.50 per share ("Common Stock"), offered pursuant to the Plan is American International Group, Inc. ("AIG" or the "Company").

Selling shareholders may from time to time reoffer or resell shares of Common Stock acquired by them under this plan directly or through agents or broker-dealers in one or more transactions on the New York Stock Exchange, at market prices and on terms prevailing at the time of the sale. See "Plan of Distribution." The Common Stock is listed on the New York Stock Exchange under the symbol "AIG."

AIG will not receive any of the proceeds from the sale of Common Stock by the persons who use this prospectus to effect resales of their shares, but will pay all costs, expenses and fees in connection with the registration of the Common Stock.

Neither the Securities and Exchange Commission nor any state securities commission has approved or disapproved of these securities or determined if this prospectus is truthful or complete. Any representation to the contrary is a criminal offense.

No dealer, sales person or other person is authorized to give any information or to represent anything not contained in this prospectus. You must not rely on any unauthorized information or representations. This prospectus is an offer to sell or to buy only the shares offered by this prospectus, but only under circumstances and in jurisdictions where it is lawful to do so. The information contained in this prospectus is current only as of the date below.

Prospectus dated December 18, 2007

#### TABLE OF CONTENTS

	Page
Incorporation by Reference of Information Concerning the Company	I-3
Selling Shareholders	I-4
Plan of Distribution	I-4
Validity of Common Stock	I-5
Experts	I-5

#### Incorporation by Reference of Information Concerning the Company

The Company files annual and quarterly reports, proxy statements and other information with the Commission. You may read and copy documents that the Company files at:

- SEC Public Reference Room
   100 F Street, N.E., Room 1580
   Washington, D.C. 20549
   Please call the Commission at 1-800-SEC-0330 for further information.
   The Company's filings are also available to the public through the Commission's web site at http://www.sec.gov.
- The New York Stock Exchange
   Broad Street
   New York, New York 10005
   The Common Stock is listed on the New York Stock Exchange.

The Commission allows the Company to "incorporate by reference" certain information the Company files with the Commission, which means that the Company can disclose important information to you by referring you to those documents. The information incorporated by reference is considered to be part of this prospectus, and later information that the Company files with the Commission will automatically update and supersede that information as well as the information included in this prospectus. The Company incorporates by reference the documents listed in the accompanying box and any future filings made with the Commission under Section 13(a), 13(c), 14 or 15(d) of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), until all the shares of Common Stock are sold, provided, however, that the Company is not incorporating by reference any information furnished (but not filed) under Item 2.02 or Item 7.01 of any Current Report on Form 8-K. This prospectus is part of a registration statement the Company filed with the Commission.

Annual Report on Form 10-K for the fiscal year ended December 31, 2006.

Quarterly Reports on Form 10-Q for the quarterly periods ended March 31, 2007, June 30, 2007 and September 30, 2007.

Current Reports on Form 8-K filed on January 19, 2007, March 1, 2007 (Item 8.01 information only), March 13, 2007, March 16, 2007, May 22, 2007, June 7, 2007 and November 15, 2007.

Current Report on Form 8-K/A filed on December 7, 2007.

The description of Common Stock contained in the registration statement on Form 8-A, dated September 20, 1984, filed pursuant to Section 12(b) of the Exchange Act.

The Company will provide without charge a copy of these filings, other than any exhibits unless the exhibits are specifically incorporated by reference into this prospectus. You may request your copy by writing or telephoning the Company at the following address:

 American International Group, Inc. Director of Investor Relations
 70 Pine Street
 New York, New York 10270
 (212) 770-7074

#### **Selling Shareholders**

Persons who are "affiliates" of AIG for purposes of the Securities Act of 1933, as amended (the "1933 Act"), and the rules promulgated by the Securities and Exchange Commission (the "Commission") thereunder, who are delivered shares of Common Stock under the Plan may use this prospectus to effect reoffers or resales of such shares. These shares of Common Stock may be deemed to be "control securities" within the meaning of General Instruction C to Form S-8.

Selling shareholders will be listed in a supplement to this prospectus.

#### **Plan of Distribution**

The selling shareholders and their pledgees, donees, transferees, distributees or other successors-in-interest may from time to time sell shares of Common Stock acquired under this Plan directly to purchasers or offer the shares of Common Stock through broker-dealers or agents. The shares of Common Stock may be sold in one or more transactions on the New York Stock Exchange at market prices and on terms prevailing at the time of the sale.

Broker-dealers or agents may receive compensation in the form of commissions, discounts or concessions from the selling shareholders in amounts to be negotiated in connection with the sales. Such broker-dealers and any other participating broker-dealers may be deemed to be "underwriters" within the meaning of the 1933 Act in connection with such sales and any such commission, discount or concession may be deemed to be underwriting discounts or commissions under the 1933 Act. In addition, any common shares covered by this prospectus that qualify for sale pursuant to Rule 144 promulgated under the 1933 Act may be sold under Rule 144 rather than pursuant to this prospectus.

Commissions and discounts, if any, attributable to the sales of the shares will be paid by the shareholders who sell the shares.

#### Validity of Common Stock

The validity of any newly issued shares of Common Stock offered hereby will be passed upon by Kathleen E. Shannon, Esq., Senior Vice President, Secretary and Deputy General Counsel of the Company. Ms. Shannon is regularly employed by the Company, participates in various employee benefit plans of the Company under which she may receive shares of Common Stock and currently beneficially owns less than 1% of the shares of outstanding Common Stock.

#### **Experts**

The consolidated financial statements, the financial statement schedules and management's assessment of the effectiveness of internal control over financial reporting (which is included in Management's Report on Internal Control over Financial Reporting) included in AIG's Annual Report on Form 10-K for the fiscal year ended December 31, 2006 (which contains an adverse opinion on the effectiveness of internal control over financial reporting), incorporated herein by reference, are so incorporated in reliance upon the report of PricewaterhouseCoopers LLP, an independent registered public accounting firm, given on the authority of that firm as experts in accounting and auditing.

#### PART II

#### INFORMATION REQUIRED IN THE REGISTRATION STATEMENT

#### Item 3. Incorporation of Documents by Reference.

The following documents have been filed by American International Group, Inc., ("AIG") with the Securities and Exchange Commission (the "Commission") (File No. 1-8787) and are incorporated herein by reference:

- (a) AIG's Annual Report on Form 10-K for the fiscal year ended December 31, 2006;
- (b) AIG's Quarterly Reports on Form 10-Q for the quarterly periods ended March 31, 2007, June 30, 2007 and September 30, 2007;
- (c) AIG's Current Reports on Form 8-K filed on January 19, 2007, March 1, 2007 (Item 8.01 information only), March 13, 2007, March 16, 2007, May 22, 2007, June 7, 2007, November 15, 2007 and December 18, 2007;
- (d) AIG's Current Report on Form 8-K/A filed on December 7, 2007; and
- (e) The description of Common Stock contained in the Registration Statement on Form 8-A, dated September 20, 1984, filed pursuant to Section 12(b) of the Securities Exchange Act of 1934, as amended (the "Exchange Act").

All documents filed by AIG after the date hereof pursuant to Section 13(a), 13(c), 14 or 15(d) of the Exchange Act, prior to the filing of a post-effective amendment which indicates that all securities offered have been sold, or which deregisters all such securities then remaining unsold, shall be deemed to be incorporated by reference in this Registration Statement and to be a part hereof from the date of filing of such documents, provided, however, that AIG is not incorporating by reference any information furnished (but not filed) under Item 2.02 or Item 7.01 of any Current Report on Form 8-K.

Any statement contained in a document incorporated or deemed incorporated by reference herein shall be deemed to be modified or superseded for purposes of this Registration Statement to the extent that a statement contained herein or in any other subsequently filed document which also is or is deemed to be incorporated by reference herein modifies or supersedes such statement. Any such statement so modified or superseded will not be deemed, except as so modified or superseded, to constitute a part of this Registration Statement.

#### Item 4. Description of Securities.

The Common Stock is registered under Section 12(b) of the Exchange Act.

Item 5. Interests of Named Experts and Counsel.

The validity of the shares of Common Stock offered pursuant to the American International Group, Inc. 2007 Amended and Restated Stock Incentive Plan (the "Plan") will be passed upon by Kathleen E. Shannon, Esq., Senior Vice President, Secretary and Deputy General Counsel of AIG. Ms. Shannon is regularly employed by AIG, participates in various AIG employee benefit plans under which she may receive shares of Common Stock and currently beneficially owns less than 1% of the outstanding shares of Common Stock.

#### Item 6. Indemnification of Directors and Officers.

The amended and restated certificate of incorporation of AIG provides that AIG shall indemnify to the full extent permitted by law any person made, or threatened to be made, a party to an action, suit or proceeding, whether civil, criminal, administrative or investigative, by reason of the fact that he or she, his or her testator or intestate is or was a director, officer or employee of AIG or serves or served any other enterprise at the request of AIG. Section 6.4 of AIG's amended and restated by-laws contains a similar provision.

The amended and restated certificate of incorporation also provides that a director will not be personally liable to AIG or its shareholders for monetary damages for breach of fiduciary duty as a director, except to the extent that the exemption from liability or limitation thereof is not permitted by the Delaware General Corporation Law.

Section 145 of the Delaware General Corporation Law permits indemnification against expenses, fines, judgments and settlements incurred by any director, officer or employee of a company in the event of pending or threatened civil, criminal, administrative or investigative proceedings, if such person was, or was threatened to be made, a party by reason of the fact that he or she is or was a director, officer or employee of the company. Section 145 also provides that the indemnification provided for therein shall not be deemed exclusive of any other rights to which those seeking indemnification may otherwise be entitled. In addition, AIG and its subsidiaries maintain a directors' and officers' liability insurance policy.

#### Item 7. Exemption From Registration Claimed.

Not applicable.

#### Item 8. Exhibits.

The exhibits are listed in the exhibit index.

## Item 9. Undertakings.

AIG hereby undertakes:

- (a) (1) To file, during any period in which offers or sales are being made, a post-effective amendment to this Registration Statement:
  - (i) To include any prospectus required by Section 10(a)(3) of the Securities Act;
  - (ii) To reflect in the prospectus any facts or events arising after the effective date of this Registration Statement (or the most recent post-effective amendment thereof) which, individually or in the aggregate, represent a fundamental change in the information set forth in this Registration Statement;

(iii) To include any material information with respect to the plan of distribution not previously disclosed in this Registration Statement or any material change to such information in this Registration Statement;

provided, however, that paragraphs (a) (1)(i) and (a) (1)(ii) do not apply if the information required to be included in a post-effective amendment by those paragraphs is contained in periodic reports filed with or furnished to the Commission by AIG pursuant to Section 13 or Section 15(d) of the Exchange Act that are incorporated by reference in this Registration Statement.

- (2) That, for the purpose of determining any liability under the Securities Act, each such post-effective amendment shall be deemed to be a new Registration Statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial *bona fide* offering thereof.
- (3) To remove from registration by means of a post-effective amendment any of the securities being registered which remain unsold at the termination of the offering.
- (b) That, for purposes of determining any liability under the Securities Act, each filing of AIG's annual report pursuant to Section 13(a) or Section 15(d) of the Exchange Act that is incorporated by reference in this Registration Statement shall be deemed to be a new Registration Statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial *bona fide* offering thereof.
- (c) Insofar as indemnification for liabilities arising under the Securities Act may be permitted to directors, officers and controlling persons of AIG pursuant to the foregoing provisions, or otherwise, AIG has been advised that in the opinion of the Commission such indemnification is against public policy as expressed in the Securities Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by AIG of expenses incurred or paid by a director, officer or controlling person of AIG in the successful defense of any action, suit or proceeding) is asserted against AIG by such director, officer or controlling person in connection with the securities being registered, AIG will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Securities Act and will be governed by the final adjudication of such issue.

#### **SIGNATURES**

Pursuant to the requirements of the Securities Act of 1933, the registrant certifies that it has reasonable grounds to believe that it meets all of the requirements for filing on Form S-8 and has duly caused this Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of New York, State of New York, on this 18 day of December, 2007.

#### AMERICAN INTERNATIONAL GROUP, INC.

By: /s/ Martin J. Sullivan

Name: Martin J. Sullivan

Title: President and Chief Executive Officer

KNOW ALL MEN BY THESE PRESENTS: that each person whose signature appears below constitutes and appoints Martin J. Sullivan and Steven J. Bensinger, and each of them severally, as true and lawful attorneys-in-fact, with full power of substitution and resubstitution, for him or her and in his or her name, place and stead, in any and all capacities to sign any and all amendments (including post-effective amendments) to this Registration Statement on Form S-8, and to file the same, with all exhibits thereto, and other documents in connection herewith, with the Securities and Exchange Commission, granting unto said attorneys-in-fact and agents, and each of them, full power and authority to do and perform each and every act and thing required and necessary to be done in and about the foregoing as fully for all intents and purposes as he or she might or could do in person, hereby ratifying and confirming all that said attorneys-in-fact and agents or any of them, or their or his substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

Pursuant to the requirements of the Securities Act of 1933, this Registration Statement has been signed by the following persons in the capacities indicated on the 18 day of December, 2007.

Signature	Title	
/s/ Martin J. Sullivan (Martin J. Sullivan)	President, Chief Executive Officer and Director (Principal Executive Officer)	
(Marian 3. Sumvun)	(Finicipal Executive Officer)	
/s/ Steven J. Bensinger	Executive Vice President and Chief Financial Officer	
(Steven J. Bensinger)	(Principal Financial Officer)	
/s/ David L. Herzog	Senior Vice President and Comptroller	
(David L. Herzog)	(Principal Accounting Officer)	
/s/ Marshall A. Cohen	Director	
(Marshall A. Cohen)		
/s/ Martin S. Feldstein (Martin S. Feldstein)	Director	
/s/ Ellen V. Futter (Ellen V. Futter)	Director	
(Ench v. Lutter)		

Signature	Title
/s/ Stephen L. Hammerman	Director
(Stephen L. Hammerman)	
/s/ Richard C. Holbrooke	Director
(Richard C. Holbrooke)	
/s/ Fred H. Langhammer	Director
(Fred H. Langhammer)	
/s/ George L. Miles, Jr.	Director
(George L. Miles, Jr.)	
/s/ Morris W. Offit	Director
(Morris W. Offit)	
/s/ James F. Orr III	Director
(James F. Orr III)	
/s/ Virginia M. Rometty	Director
(Virginia M. Rometty)	
/s/ Michael H. Sutton	Director
(Michael H. Sutton)	
/s/ Edmund S.W. Tse	Director
(Edmund S.W. Tse)	
/s/ Robert B. Willumstad	Director
(Robert B. Willumstad)	
/s/ Frank G. Zarb	Director
(Frank G. Zarb)	

## EXHIBIT INDEX

Exhibit Number	Description	Location
4	American International Group, Inc. Amended and Restated 2007 Stock Incentive Plan	Filed as exhibit hereto.
5	Opinion of Kathleen E. Shannon re validity	Filed as exhibit hereto.
10	Material contracts	
	(a) Form of Stock Option Award Agreement under the AIG Amended and Restated 2007 Stock Incentive Plan.	Filed as exhibit hereto.
	(b) Form of Performance RSU Award Agreement under the AIG Amended and Restated 2007 Stock Incentive Plan	Filed as exhibit hereto.
	(c) Form of Time-Vested RSU Award Agreement under the AIG Amended and Restated 2007 Stock Incentive Plan	Filed as exhibit hereto.
	(d) Form of Time-Vested RSU Award Agreement with four- year pro rata vesting under the AIG Amended and Restated 2007 Stock Incentive Plan	Filed as exhibit hereto.
	(e) Form of Time-Vested RSU Award Agreement with three- year pro rata vesting under the AIG Amended and Restated 2007 Stock Incentive Plan	Filed as exhibit hereto.
	(f) Form of Non-Employee Director Deferred Stock Units Award Agreement under the AIG Amended and Restated 2007 Stock Incentive Plan	Filed as exhibit hereto.
15	Letter re unaudited interim financial information	Not applicable.

Exhibit Number	Description	Location
23	Consents of experts and counsel	
	(a) PricewaterhouseCoopers LLP.	Filed as exhibit hereto.
	(b) Kathleen E. Shannon, Esq.	Included in Exhibit 5.
24	Power of Attorney	Included in signature pages.

## AMERICAN INTERNATIONAL GROUP, INC. AMENDED AND RESTATED 2007 STOCK INCENTIVE PLAN

## AMERICAN INTERNATIONAL GROUP, INC. Amended and Restated 2007 Stock Incentive Plan

## **Table of Contents**

ARTICLE I—GENERAL	
1.1 Purpose	3
1.2 Definitions of Certain Terms	3
1.3 Administration	5
1.4 Persons Eligible for Awards	$\epsilon$
1.5 Types of Awards Under Plan	$\epsilon$
1.6 Shares of Common Stock Available for Awards	6
ARTICLE II—AWARDS UNDER THE PLAN	
2.1 Agreements Evidencing Awards	7
2.2 No Rights as a Shareholder	3
2.3 Stock Options	3
2.4 Stock Appreciation Rights	g
2.5 Restricted Shares	10
2.6 Restricted Stock Units	10
2.7 Dividend Equivalent Rights	11
2.8 Other Stock-Based Awards	11
2.9 Certain Restrictions	11
ARTICLE III—MISCELLANEOUS	
3.1 Amendment of the Plan	11
3.2 Tax Withholding	12
3.3 Required Consents and Legends	12
3.4 Right of Offset	13
3.5 Nonassignability; No Hedging	13
3.6 Successor Entity	13
3.7 Right of Discharge Reserved	13
3.8 Nature of Payments	13
3.9 Non-Uniform Determinations	14
3.10 Other Payments or Awards	14
3.11 Plan Headings	14
3.12 Termination of Plan	14
3.13 Section 409A Payment Delay	15
3.14 Governing Law	15
3.15 Severability; Entire Agreement	15
3.16 Waiver of Claims	15
3.17 No Third Party Beneficiaries	15
3.18 Successors and Assigns of AIG	16
3.19 Date of Adoption and Approval of Shareholders	16

#### AMERICAN INTERNATIONAL GROUP, INC. Amended and Restated 2007 Stock Incentive Plan

#### ARTICLE I—GENERAL

#### 1.1 Purpose

The purpose of the American International Group, Inc. Amended and Restated 2007 Stock Incentive Plan is to attract, retain and motivate officers, directors and key employees of American International Group, Inc. and its consolidated subsidiaries, to compensate them for their contributions to the long-term growth and profits of the Company and to encourage them to acquire a proprietary interest in the success of the Company.

This 2007 Stock Incentive Plan replaces the American International Group, Inc. Amended and Restated 1999 Stock Option Plan (as amended to the Effective Date, the "Stock Option Plan"), the American International Group, Inc. Amended and Restated 2002 Stock Incentive Plan (as amended to the Effective Date, the "SIP") and the American International Group, Inc. Director Stock Plan (as amended to the Effective Date, the "Director Plan") for Awards granted on or after the Effective Date. Awards may not be granted under any of the Stock Option Plan, the SIP or the Director Plan beginning on the Effective Date, but this 2007 Stock Incentive Plan will not affect the terms or conditions of any stock option, restricted stock unit or other award made under the Stock Option Plan, the SIP or the Director Plan before the Effective Date.

#### 1.2 Definitions of Certain Terms

For purposes of this 2007 Stock Incentive Plan, the following terms have the meanings set forth below:

- "AIG" means American International Group, Inc. or a successor entity contemplated by Section 3.6.
- "Assurance Agreement" means the Assurance Agreement, by AIG in favor of eligible employees dated as of June 27, 2005, relating to certain obligations of Starr International Company, Inc. (as such agreement may be amended, supplemented, extended, modified or replaced from time to time).
  - "Award" means an award made pursuant to the Plan.
- "Award Agreement" means the written document by which each Award is evidenced, and which may, but need not be (as determined by the Committee) executed or acknowledged by a Grantee as a condition to receiving an Award or the benefits under an Award, and which sets forth the terms and provisions applicable to Awards granted under the Plan to such Grantee.
  - "Board" means the Board of Directors of AIG.
  - "Certificate" means a stock certificate (or other appropriate document or evidence of ownership) representing shares of Common Stock.

- "Code" means the Internal Revenue Code of 1986, as amended from time to time, or any successor thereto, and the applicable rulings and regulations thereunder.
  - "Committee" has the meaning set forth in Section 1.3.1.
- "Common Stock" means the common stock of AIG, par value \$2.50 per share, and any other securities or property issued in exchange therefor or in lieu thereof pursuant to Section 1.6.4.
  - "Company" means AIG and its consolidated subsidiaries.
  - "Consent" has the meaning set forth in Section 3.3.2.
  - "Covered Person" has the meaning set forth in Section 1.3.3.
  - "Director" means a member of the Board or a member of the board of directors of a consolidated subsidiary of AIG.
  - "Effective Date" means May 16, 2007.
  - "Employee" means a regular, active employee of the Company.
- "Employment" means a Grantee's performance of services for the Company, as an Employee, as determined by the Committee. The terms "employ" and "employed" will have their correlative meanings.
- "Exchange Act" means the Securities Exchange Act of 1934, as amended from time to time, or any successor thereto, and the applicable rules and regulations thereunder.
- "Fair Market Value" means, with respect to a share of Common Stock on any day, the fair market value as determined in accordance with a valuation methodology approved by the Committee, unless determined as otherwise specified herein.
  - "Grantee" means an Employee or Director who receives an Award.
- "Incentive Stock Option" means an option to purchase shares of Common Stock that is intended to be designated as an "incentive stock option" within the meaning of Sections 421 and 422 of the Code, as now constituted or subsequently amended, or pursuant to a successor of the Code, and which is designated as an Incentive Stock Option in the applicable Award Agreement.
- "Non-Employee Director" means a Director who is a "non-employee director" within the meaning of Rule 16b-3(b)(3) under the Exchange Act or any successor thereto.
  - "Officer" means an Employee who is an "officer" within the meaning of Rule 16a-1(f) under the Exchange Act.
  - "Plan" means this American International Group, Inc. Amended and Restated 2007 Stock Incentive Plan, as amended from time to time.
  - "Plan Action" will have the meaning set forth in Section 3.3.1.

"Securities Act" means the Securities Act of 1933, as amended from time to time, or any successor thereto, and the applicable rules and regulations thereunder.

#### 1.3 Administration

1.3.1 The Compensation and Management Resources Committee of the Board (as constituted from time to time, and including any successor committee, the "Committee") will administer the Plan. The members of the Committee will be drawn solely from such members of the Board who are not and have not been Officers of the Company. The Committee is authorized, subject to the provisions of the Plan, to establish such rules and regulations as it deems necessary for the proper administration of the Plan and to make such determinations and interpretations and to take such action in connection with the Plan and any Award granted thereunder as it deems necessary or advisable. All determinations and interpretations made by the Committee will be final, binding and conclusive on all Grantees and on their legal representatives and beneficiaries. The Committee will have the authority, in its absolute discretion, to determine the persons who will receive Awards, the time when Awards will be granted, the terms of such Awards and the number of shares of Common Stock, if any, which will be subject to such Awards. Unless otherwise provided in an Award Agreement, the Committee reserves the authority, in its absolute discretion,(a) in the case of death, disability or retirement (either before or at normal retirement age), to amend any outstanding Award Agreement in any respect, whether or not the rights of the Grantee of such Award are adversely affected (but subject to Sections 2.3.6, 2.4.5 and 2.9), including, without limitation, to accelerate the time or times at which the Award becomes vested, unrestricted or may be exercised, to waive or amend any restrictions or conditions set forth in such Award Agreement, or to impose new restrictions and conditions, or to reflect a change in the Grantee's circumstances and (b) to determine whether, to what extent and under what circumstances and method or methods (i) Awards may be (A) settled in shares of Common Stock, other securities, other Awards or other property or (B) canceled, forfeited or suspended, (ii) shares of Common Stock, other securities, other Awards or other property, and other amounts payable with respect to an Award may be deferred either automatically or at the election of the Grantee thereof or of the Committee and (iii) Awards may be settled by the Company or any of its designees. Notwithstanding anything to the contrary contained herein, the Board may, in its sole discretion, at any time and from time to time, grant Awards (other than grants to Directors) or administer the Plan, in which case the Board will have all of the authority and responsibility granted to the Committee herein. If so determined by the Committee, any Award made to an Officer will be made by the full Board or a committee or subcommittee of the Board composed of at least two Non-Employee Directors of AIG.

1.3.2 Actions of the Committee may be taken by the vote of a majority of its members. To the extent not inconsistent with applicable law and the rules and regulations of the New York Stock Exchange, (a) the Committee may delegate any of its powers under the Plan to a subcommittee of the Committee or to one of its members, (b) the Committee may allocate among its members any of its administrative responsibilities and (c) notwithstanding anything to the contrary contained herein, the Committee may delegate the determination of Awards (and related administrative responsibilities) to Employees who are not Officers to one or more officers of AIG designated by the Committee from time to time.

1.3.3 No Director or Employee (each such person, a "Covered Person") will have any liability to any person (including any Grantee) for any action taken or omitted to be taken or any determination made in good faith with respect to the Plan or any Award. Each Covered Person will be indemnified and held harmless by AIG against and from any loss, cost, liability or expense (including attorneys' fees) that may be imposed upon or incurred by such Covered Person in connection with or resulting from any action, suit or proceeding to which such Covered Person may be a party or in which such Covered Person may be involved by reason of any action taken or omitted to be taken under the Plan or any Award Agreement and against and from any and all amounts paid by such Covered Person, with AIG's approval, in settlement thereof, or paid by such Covered Person in satisfaction of any judgment in any such action, suit or proceeding against such Covered Person, provided that AIG will have the right, at its own expense, to assume and defend any such action, suit or proceeding and, once AIG gives notice of its intent to assume the defense, AIG will have sole control over such defense with counsel of AIG's choice. The foregoing right of indemnification will not be available to a Covered Person to the extent that a court of competent jurisdiction in a final judgment or other final adjudication, in either case, not subject to further appeal, determines that the acts or omissions of such Covered Person giving rise to the indemnification claim resulted from such Covered Person's bad faith, fraud or willful misconduct. The foregoing right of indemnification will not be exclusive of any other rights of indemnification to which Covered Persons may be entitled under AIG's Restated Certificate of Incorporation or By-laws, as a matter of law, or otherwise, or any other power that AIG may have to indemnify such persons or hold them harmless.

#### 1.4 Persons Eligible for Awards

Awards under the Plan may be made to Employees and Directors.

#### 1.5 Types of Awards Under Plan

Awards may be made under the Plan in the form of any of the following, in each case in respect of Common Stock: (a) stock options, (b) stock appreciation rights, (c) restricted shares, (d) restricted stock units, (e) dividend equivalent rights and (f) other equity-based or equity-related Awards that the Committee determines to be consistent with the purposes of the Plan and the interests of the Company.

#### 1.6 Shares of Common Stock Available for Awards

1.6.1 *Common Stock Subject to the Plan.* Subject to the other provisions of this Section 1.6, the total number of shares of Common Stock that may be granted under the Plan is 180,000,000. Such shares of Common Stock may, in the discretion of the Committee, be either authorized but unissued shares or shares previously issued and reacquired by AIG.

1.6.2 *Share Counting*. The number of shares of Common Stock granted under the Plan per year will be determined as follows: (a) each stock option, stock appreciate right and similar Award will count as 1 share of Common Stock and (b) each restricted share, restricted stock unit and similar Award will count as 2.9 shares of Common Stock. Shares of Common Stock issued in connection with awards that are

assumed, converted or substituted as a result of the Company's acquisition of another company (including by way of merger, combination or similar transaction) will not count against the number of shares that may be issued under the Plan.

1.6.3 Replacement of Shares. If any Award is forfeited, expires, terminates or otherwise lapses, in whole or in part, without the delivery of Common Stock, then the shares of Common Stock covered by such forfeited, expired, terminated or lapsed award (counted in accordance with Section 1.6.2) will again be available for grant under the Plan. In addition, the following will be added to the number of shares available for grant under the Plan: (1) the number of shares of Common Stock underlying awards granted and outstanding under the Stock Option Plan before the Effective Date that are forfeited, expire, terminate or otherwise lapse on or after the Effective Date, in whole or in part, without the delivery of Common Stock and (2) the number of shares of Common Stock underlying awards granted and outstanding under the SIP before the Effective Date that are forfeited, expire, terminate or otherwise lapse on or after the Effective Date, in whole or in part, without the delivery of Common Stock (in each case, counted in accordance with Section 1.6.2). For the avoidance of doubt, the following shall not again become available for issuance under the Plan: (A) any shares of Common Stock withheld in respect of taxes, (B) any shares tendered or withheld to pay the exercise price of stock options, (C) any shares repurchased by the Company from the optionee with the proceeds from the exercise of stock options and (D) any shares subject to stock appreciation rights but not issued on exercise as a result of the operation of Section 2.4.4.

1.6.4 Adjustments. The Committee will adjust the number of shares of Common Stock authorized pursuant to Section 1.6.1 and adjust equitably the terms of any outstanding Awards (including, without limitation, the number of shares of Common Stock covered by each outstanding Award, the type of property to which the Award is subject and the exercise or strike price of any Award), in such manner as it deems appropriate (including, without limitation, by payment of cash) to preserve the benefits or potential benefits intended to be made available to grantees of Awards, for any increase or decrease in the number of issued shares of Common Stock resulting from a recapitalization, stock split, stock dividend, combination or exchange of shares of Common Stock, merger, consolidation, rights offering, separation, reorganization or liquidation, or any other change in the corporate structure or shares of AIG. After any adjustment made pursuant to this Section 1.6.4, the number of shares of Common Stock subject to each outstanding Award will be rounded down to the nearest whole number.

#### ARTICLE II—AWARDS UNDER THE PLAN

#### 2.1 Agreements Evidencing Awards

Each Award granted under the Plan will be evidenced by an Award Agreement that will contain such provisions and conditions as the Committee deems appropriate. Unless otherwise provided herein, the Committee may grant Awards in tandem with or in substitution for any other Award or Awards granted under the Plan or any award granted under any other plan of AIG. By accepting an Award pursuant to the Plan, a Grantee thereby agrees that the Award will be subject to all of the terms and provisions of the Plan and the applicable Award Agreement.

#### 2.2 No Rights as a Shareholder

No Grantee (or other person having rights pursuant to an Award) shall have any of the rights of a shareholder of AIG with respect to shares of Common Stock subject to an Award until the delivery of such shares. Except as otherwise provided in Section 1.6.4, no adjustments will be made for dividends, distributions or other rights (whether ordinary or extraordinary, and whether in cash, Common Stock, other securities or other property) for which the record date is before the date the Certificates for the shares are delivered.

#### 2.3 Options

- 2.3.1 *Grant.* Stock options may be granted to eligible recipients in such number and at such times during the term of the Plan as the Committee or the Board may determine; *provided, however*, that the maximum number of shares of Common Stock as to which stock options may be granted under the Plan to any one individual in any one year may not exceed 1,000,000 shares (as adjusted pursuant to the provisions of Section 1.6.4).
- 2.3.2 *Incentive Stock Options*. At the time of grant, the Committee will determine (a) whether all or any part of a stock option granted to an eligible employee will be an Incentive Stock Option and (b) the number of shares subject to such Incentive Stock Option; *provided*, *however*, that (1) the aggregate fair market value (determined as of the time the option is granted) of the stock with respect to which Incentive Stock Options are exercisable for the first time by an eligible employee during any calendar year (under all such plans of AIG and of any subsidiary corporation of AIG) will not exceed \$100,000 and (2) no Incentive Stock Option (other than an Incentive Stock Option that may be assumed or issued by the Company in connection with a transaction to which Section 424(a) of the Code applies) may be granted to a person who is not eligible to receive an Incentive Stock Option under the Code. The form of any stock option which is entirely or in part an Incentive Stock Option will clearly indicate that such stock option is an Incentive Stock Option or, if applicable, the number of shares subject to the Incentive Stock Option.
- 2.3.3 *Exercise Price*. The exercise price per share with respect to each stock option will be determined by the Committee but will not be less than the Fair Market Value of the Common Stock. Unless otherwise noted in the Award Agreement, the Fair Market Value of the Common Stock will be its closing price on the New York Stock Exchange on the date of grant of the Award of stock options.
- 2.3.4 *Term of Stock Option*. In no event will any stock option be exercisable after the expiration of 10 years from the date on which the stock option is granted.
- 2.3.5 Exercise of Stock Option and Payment for Shares. The shares of Common Stock covered by each stock option may not be purchased for one year after the date on which the stock option is granted (except in the case of termination of Employment due to death, disability or retirement), but thereafter may be purchased in such installments as will be determined in the Award Agreement at the time the stock option is granted. Subject to any limitations in the applicable Award Agreement, any shares not purchased on the applicable installment date may be purchased thereafter at any time before the final expiration of the stock option. To exercise a stock option, the Grantee must give written notice to AIG specifying the

number of shares to be purchased and accompanied by payment of the full purchase price therefor in cash or by certified or official bank check or in another form as determined by the Company, including: (a) personal check, (b) shares of Common Stock, valued as of the exercise date, of the same class as those to be granted by exercise of the stock option, (c) any other form of consideration approved by the Company and permitted by applicable law and (d) any combination of the foregoing. Any person exercising a stock option will make such representations and agreements and furnish such information as the Committee may in its discretion deem necessary or desirable to assure compliance by AIG, on terms acceptable to AIG, with the provisions of the Securities Act and any other applicable legal requirements. If a Grantee so requests, shares purchased may be issued in the name of the Grantee and another jointly with the right of survivorship.

2.3.6 *Repricing*. Except as otherwise permitted by Section 1.6.4, reducing the exercise price of stock options issued and outstanding under the Plan, including through amendment, cancellation in exchange for the grant of a substitute Award or repurchase for cash or other consideration (in each case that has the effect of reducing the exercise price), will require approval of the shareholders.

#### 2.4 Stock Appreciation Rights

- 2.4.1 *Grant.* Stock appreciation rights may be granted to eligible recipients in such number and at such times during the term of the Plan as the Committee or the Board may determine; *provided*, *however*, that the maximum number of shares of Common Stock as to which stock appreciation rights may be granted under the Plan to any one individual in any one year may not exceed 1,000,000 shares (as adjusted pursuant to the provisions of Section 1.6.4).
- 2.4.2 *Exercise Price*. The exercise price per share with respect to each stock appreciation right will be determined by the Committee but will not be less than the Fair Market Value of the Common Stock. Unless otherwise noted in the Award Agreement, the Fair Market Value of the Common Stock will be its closing price on the New York Stock Exchange on the date of grant of the Award of stock appreciation rights.
- 2.4.3 *Term of Stock Appreciation Right*. In no event will any stock appreciation right be exercisable after the expiration of 10 years from the date on which the Stock Appreciation Right is granted.
- 2.4.4 Exercise of Stock Appreciation Right and Delivery of Shares. Each stock appreciation right may not be exercised for one year after the date on which the stock appreciation right is granted (except in the case of termination of Employment due to death, disability or retirement), but thereafter may be exercised in such installments as may be determined in the Award Agreement at the time the stock appreciation right is granted. Subject to any limitations in the applicable Award Agreement, any stock appreciation rights not exercised on the applicable installment date may be exercised thereafter at any time before the final expiration of the stock appreciation right. To exercise a stock appreciation right, the Grantee must give written notice to AIG specifying the number of stock appreciation rights to be exercised. Upon exercise of stock appreciation rights, shares of Common Stock with a Fair Market Value equal to (a) the excess of (1) the Fair Market Value of the Common Stock on the date of exercise over (2) the exercise price of such stock

appreciation right *multiplied by* (b) the number of stock appreciation rights exercised will be delivered to the Grantee. Any person exercising a stock appreciation right will make such representations and agreements and furnish such information as the Committee may in its discretion deem necessary or desirable to assure compliance by AIG, on terms acceptable to AIG, with the provisions of the Securities Act and any other applicable legal requirements. If a Grantee so requests, shares purchased may be issued in the name of the Grantee and another jointly with the right of survivorship.

2.4.5 *Repricing*. Except as otherwise permitted by Section 1.6.4, reducing the exercise price of stock appreciation rights issued and outstanding under the Plan, including through amendment, cancellation in exchange for the grant of a substitute Award or repurchase for cash or other consideration (in each case that has the effect of reducing the exercise price), will require approval of the shareholders.

#### 2.5 Restricted Shares

2.5.1 *Grants*. The Committee may grant or offer for sale restricted shares in such amounts and subject to Section 2.9 and such terms and conditions as the Committee may determine. Upon the delivery of such shares, the Grantee will have the rights of a shareholder with respect to the restricted shares, subject to Section 2.9 and any other restrictions and conditions as the Committee may include in the applicable Award Agreement. In the event that a Certificate is issued in respect of restricted shares, such Certificate may be registered in the name of the Grantee but will be held by AIG or its designated agent until the time the restrictions lapse.

2.5.2 Right to Vote and Receive Dividends on Restricted Shares. Each Grantee of an Award of restricted shares will, during the period of restriction, be the beneficial and record owner of such restricted shares and will have full voting rights with respect thereto. Unless the Committee determines otherwise in an Award Agreement, during the period of restriction, all ordinary cash dividends (as determined by the Committee in its sole discretion) paid upon any restricted share will be retained by the Company for the account of the relevant Grantee. Such dividends will revert back to the Company if for any reason the restricted share upon which such dividends were paid reverts back to the Company. Upon the expiration of the period of restriction, all such dividends made on such restricted share and retained by the Company will be paid to the relevant Grantee. Unless the applicable Award Agreement provides otherwise, additional shares or other property distributed to the Grantee in respect of restricted shares, as dividends or otherwise, will be subject to the same restrictions applicable to such restricted shares.

#### 2.6 Restricted Stock Units

The Committee may grant Awards of restricted stock units in such amounts and subject to Section 2.9 and such terms and conditions as the Committee may determine. A Grantee of a restricted stock unit will have only the rights of a general unsecured creditor of AIG until delivery of shares of Common Stock or other securities or property is made as specified in the applicable Award Agreement. On the delivery date specified in the Award Agreement, the Grantee of each restricted stock unit not previously forfeited or terminated will receive one share of Common

Stock, or securities or other property equal in value to a share of Common Stock or a combination thereof, as specified by the Committee.

#### 2.7 Dividend Equivalent Rights

The Committee may include in the Award Agreement with respect to any Award a dividend equivalent right entitling the Grantee to receive amounts equal to all or any portion of the dividends that would be paid on the shares of Common Stock covered by such Award if such shares had been delivered pursuant to such Award. The grantee of a dividend equivalent right will have only the rights of a general unsecured creditor of AIG until payment of such amounts is made as specified in the applicable Award Agreement. In the event such a provision is included in an Award Agreement, the Committee will determine whether such payments will be made in cash, in shares of Common Stock or in another form, whether they will be conditioned upon the exercise of the Award to which they relate, the time or times at which they will be made, and such other terms and conditions as the Committee will deem appropriate.

#### 2.8 Other Stock-Based Awards

Subject to Section 2.9, the Committee may grant other types of equity-based or equity-related Awards (including the grant or offer for sale of unrestricted shares of Common Stock) in such amounts and subject to such terms and conditions as the Committee may determine. Such Awards may entail the transfer of actual shares of Common Stock to Award recipients and may include Awards designed to comply with or take advantage of the applicable local laws of jurisdictions other than the United States. Notwithstanding anything to the contrary contained herein, a maximum of 3,000,000 shares of Common Stock (which, for purposes of Section 1.6, would represent 8,700,000 of the available shares of Common Stock under this Plan when counted in accordance with Section 1.6.2) may be delivered under the Plan pursuant to this Section 2.8 without giving effect to the restrictions set forth in Section 2.9.

#### 2.9 Certain Restrictions

In the case of an Award in the form of restricted shares, restricted stock units or similar Awards, at least three years must elapse before the delivery or payment of shares of Common Stock or other property, *except* in the case of (a) termination of Employment due to death, disability or retirement, or (b) an Award that the Committee determines is performance based, in which case at least one year must elapse. To the extent shares under the Plan are delivered by AIG to perform its obligations under the Assurance Agreement, the period referred to in the preceding sentence will be determined from the date of the underlying award by Starr International Company, Inc.

#### ARTICLE III—MISCELLANEOUS

#### 3.1 Amendment of the Plan

3.1.1 Unless otherwise provided in an Award Agreement, the Board may from time to time suspend, discontinue, revise or amend the Plan in any respect

whatsoever, including in any manner that adversely affects the rights, duties or obligations of any Grantee of an Award.

3.1.2 Unless otherwise determined by the Board, shareholder approval of any suspension, discontinuance, revision or amendment will be obtained only to the extent necessary to comply with any applicable laws, regulations or rules of a securities exchange or self-regulatory agency.

#### 3.2 Tax Withholding

As a condition to the delivery of any shares of Common Stock pursuant to any Award or the lifting or lapse of restrictions on any Award, or in connection with any other event that gives rise to a federal or other governmental tax withholding obligation on the part of the Company relating to an Award (including, without limitation, FICA tax), (a) the Company may deduct or withhold (or cause to be deducted or withheld) from any payment or distribution to a Grantee whether or not pursuant to the Plan (including shares of Common Stock otherwise deliverable) or (b) the Committee will be entitled to require that the Grantee remit cash to the Company (through payroll deduction or otherwise), in each case in an amount sufficient in the opinion of the Company to satisfy such withholding obligation.

#### 3.3 Required Consents and Legends

3.3.1 If the Committee at any time determines that any Consent (as hereinafter defined) is necessary or desirable as a condition of, or in connection with, the granting of any Award, the delivery of shares of Common Stock or the delivery of any securities or other property under the Plan, or the taking of any other action thereunder (each such action a "Plan Action"), then such Plan Action will not be taken, in whole or in part, unless and until such Consent will have been effected or obtained to the full satisfaction of the Committee. The Committee may direct that any Certificate evidencing shares delivered pursuant to the Plan will bear a legend setting forth such restrictions on transferability as the Committee may determine to be necessary or desirable, and may advise the transfer agent to place a stop transfer order against any legended shares.

3.3.2 The term "Consent" as used in this Article III with respect to any Plan Action includes (a) any and all listings, registrations or qualifications in respect thereof upon any securities exchange or under any federal, state, or local law, or law, rule or regulation of a jurisdiction outside the United States, (b) or any other matter, which the Committee may deem necessary or desirable to comply with the terms of any such listing, registration or qualification or to obtain an exemption from the requirement that any such listing, qualification or registration be made, (c) any and all other consents, clearances and approvals in respect of a Plan Action by any governmental or other regulatory body or any stock exchange or self-regulatory agency and (d) any and all consents required by the Committee. Nothing herein will require the Company to list, register or qualify the shares of Common Stock on any securities exchange.

#### 3.4 Right of Offset

The Company will have the right to offset against its obligation to deliver shares of Common Stock (or other property) under the Plan or any Award Agreement any outstanding amounts (including, without limitation, travel and entertainment or advance account balances, loans, repayment obligations under any Awards, or amounts repayable to the Company pursuant to tax equalization, housing, automobile or other employee programs) that the Grantee then owes to the Company and any amounts the Committee otherwise deems appropriate pursuant to any tax equalization policy or agreement.

#### 3.5 Nonassignability; No Hedging

No Award (or any rights and obligations thereunder) granted to any person under the Plan may be sold, exchanged, transferred, assigned, pledged, hypothecated or otherwise disposed of or hedged, in any manner (including through the use of any cash-settled instrument), whether voluntarily or involuntarily and whether by operation of law or otherwise, other than by will or by the laws of descent and distribution. Any sale, exchange, transfer, assignment, pledge, hypothecation, or other disposition in violation of the provisions of this Section 3.5 will be null and void and any Award which is hedged in any manner will immediately be forfeited. All of the terms and conditions of the Plan and the Award Agreements will be binding upon any permitted successors and assigns.

#### 3.6 Successor Entity

Unless otherwise provided in the applicable Award Agreement and except as otherwise determined by the Committee, in the event of a merger, consolidation, mandatory share exchange or other similar business combination of AIG with or into any other entity ("Successor Entity") or any transaction in which another person or entity acquires all of the issued and outstanding Common Stock of AIG, or all or substantially all of the assets of AIG, outstanding Awards may be assumed or a substantially equivalent award may be substituted by such successor entity or a parent or subsidiary of such successor entity.

#### 3.7 Right of Discharge Reserved

Nothing in the Plan or in any Award Agreement will confer upon any Grantee the right to continued Employment by the Company or affect any right which the Company may have to terminate such Employment.

#### 3.8 Nature of Payments

3.8.1 Any and all grants of Awards and deliveries of Common Stock, securities or other property under the Plan will be in consideration of services performed or to be performed for the Company by the Grantee. Awards under the Plan may, in the discretion of the Committee, be made in substitution in whole or in part for cash or other compensation otherwise payable to a participant in the Plan. Only whole shares of Common Stock will be delivered under the Plan. Awards will, to the extent reasonably practicable, be aggregated in order to eliminate any fractional shares.

Fractional shares may, in the discretion of the Committee, be forfeited or be settled in cash or otherwise as the Committee may determine.

3.8.2 All such grants and deliveries will constitute a special discretionary incentive payment to the Grantee and will not be required to be taken into account in computing the amount of salary or compensation of the Grantee for the purpose of determining any contributions to or any benefits under any pension, retirement, profit-sharing, bonus, life insurance, severance or other benefit plan of the Company or under any agreement with the Grantee, unless the Company specifically provides otherwise.

#### 3.9 Non-Uniform Determinations

3.9.1 The Committee's determinations under the Plan and Award Agreements need not be uniform and may be made by it selectively among persons who receive, or are eligible to receive, Awards under the Plan (whether or not such persons are similarly situated). Without limiting the generality of the foregoing, the Committee will be entitled, among other things, to make non-uniform and selective determinations under Award Agreements, and to enter into non-uniform and selective Award Agreements, as to (a) the persons to receive Awards, (b) the terms and provisions of Awards and (c) whether a Grantee's Employment has been terminated for purposes of the Plan.

3.9.2 To the extent the Committee deems it necessary, appropriate or desirable to comply with foreign law or practices and to further the purposes of the Plan, the Committee may, without amending the Plan, establish special rules applicable to Awards to Grantees who are foreign nationals, are employed outside the United States or both and grant Awards (or amend existing Awards) in accordance with those rules.

#### 3.10 Other Payments or Awards

Nothing contained in the Plan will be deemed in any way to limit or restrict the Company from making any award or payment to any person under any other plan, arrangement or understanding, whether now existing or hereafter in effect.

#### 3.11 Plan Headings

The headings in the Plan are for the purpose of convenience only and are not intended to define or limit the construction of the provisions hereof.

#### 3.12 Termination of Plan

The Board reserves the right to terminate the Plan at any time; *provided*, *however*, that in any case, the Plan will terminate May 16, 2017, and *provided further*, that all Awards made under the Plan before its termination will remain in effect until such Awards have been satisfied or terminated in accordance with the terms and provisions of the Plan and the applicable Award Agreements.

#### 3.13 Section 409A Payment Delay

Notwithstanding any provision to the contrary herein, to the extent any payment to be made pursuant to an Award in connection with a termination of an Employee's Employment would be subject to the additional tax of Section 409A of the Code, the payment will be delayed until six months after such a termination (or earlier death or disability (within the meaning of Section 409A of the Code)).

#### 3.14 Governing Law

THE PLAN WILL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK, WITHOUT REGARD TO PRINCIPLES OF CONFLICT OF LAWS.

#### 3.15 Severability; Entire Agreement

If any of the provisions of the Plan or any Award Agreement is finally held to be invalid, illegal or unenforceable (whether in whole or in part), such provision will be deemed modified to the extent, but only to the extent, of such invalidity, illegality or unenforceability and the remaining provisions will not be affected thereby; provided that if any of such provisions is finally held to be invalid, illegal, or unenforceable because it exceeds the maximum scope determined to be acceptable to permit such provision to be enforceable, such provision will be deemed to be modified to the minimum extent necessary to modify such scope in order to make such provision enforceable hereunder. The Plan and any Award Agreements contain the entire agreement of the parties with respect to the subject matter thereof and supersede all prior agreements, promises, covenants, arrangements, communications, representations and warranties between them, whether written or oral with respect to the subject matter thereof.

#### 3.16 Waiver of Claims

Each Grantee of an Award recognizes and agrees that before being selected by the Committee to receive an Award he or she has no right to any benefits hereunder. Accordingly, in consideration of the Grantee's receipt of any Award hereunder, he or she expressly waives any right to contest the amount of any Award, the terms of any Award Agreement, any determination, action or omission hereunder or under any Award Agreement by the Committee, the Company or the Board, or any amendment to the Plan or any Award Agreement (other than an amendment to the Plan or an Award Agreement to which his or her consent is expressly required by the express terms of an Award Agreement).

#### 3.17 No Third Party Beneficiaries

Except as expressly provided therein, neither the Plan nor any Award Agreement will confer on any person other than the Company and the Grantee of any Award any rights or remedies thereunder. The exculpation and indemnification provisions of Section 1.3.2 will inure to the benefit of a Covered Person's estate and beneficiaries and legatees.

## 3.18 Successors and Assigns of AIG

The terms of the Plan will be binding upon and inure to the benefit of AIG and any successor entity contemplated by Section 3.6.

## 3.19 Date of Adoption and Approval of Shareholders

The Plan was adopted on March 14, 2007 by the Board and approved by the shareholders of AIG at the 2007 Annual Meeting of Shareholders. The Plan was amended and restated by the Board on November 14, 2007.

 $[Letterhead\ of\ American\ International\ Group,\ Inc.]$ 

December 18, 2007

American International Group, Inc. 70 Pine Street New York, New York 10270

#### Ladies and Gentlemen:

In connection with the registration under the Securities Act of 1933, as amended (the "Act"), of 135,000,000 shares (the "Securities") of Common Stock, par value \$2.50 per share (the "Common Stock"), of American International Group, Inc., a Delaware corporation (the "Company"), I, as Senior Vice President, Secretary and Deputy General Counsel of the Company, have examined such corporate records, certificates and other documents, and such questions of law, as I have considered necessary or appropriate for the purposes of this opinion.

Upon the basis of such examination, I advise you that, in my opinion, with respect the Securities to be issued under the American International Group, Inc. Amended and Restated 2007 Stock Incentive Plan (the "Plan"), when the registration statement relating to the Securities (the "Registration Statement") has become effective under the Act and the Securities are duly issued and delivered in accordance with the terms of the Plan and the terms and conditions of any award agreement under the Plan, the Securities will be validly issued, fully paid and non-assessable.

The foregoing opinion is limited to the Federal laws of the United States and the General Corporation Law of the State of Delaware, and I am expressing no opinion as to the effect of the laws of any other jurisdiction.

With your approval, I have relied as to certain matters on information obtained from public officials, officers of the Company and other sources believed by me to be responsible, and I have assumed that the certificates for the Securities will conform to the specimen thereof examined by me and will be duly countersigned by a transfer agent and duly registered by a registrar of the Common Stock, and that the signatures on all documents examined by me are genuine, assumptions that I have not independently verified.

I hereby consent to the filing of this opinion as an exhibit to the Registration Statement. In giving such consent, I do not thereby admit that I am in the category of persons whose consent is required under Section 7 of the Act.

Very truly yours,

/s/ Kathleen E. Shannon
Kathleen E. Shannon
Senior Vice President, Secretary and
Deputy General Counsel

## AMERICAN INTERNATIONAL GROUP, INC. AMENDED AND RESTATED 2007 STOCK INCENTIVE PLAN STOCK OPTION AWARD AGREEMENT

This award agreement (this "Award Agreement") sets forth the terms and conditions of an award (this "Award") of stock options ("Options") to purchase shares of Common Stock ("Shares") granted to you under the American International Group, Inc. Amended and Restated 2007 Stock Incentive Plan (the "Plan").

- 1. <u>The Plan</u>. This Award is made pursuant to the Plan, the terms of which are incorporated in this Award Agreement. Capitalized terms used in this Award Agreement that are not defined in this Award Agreement, or in the attached Glossary of Terms, have the meanings as used or defined in the Plan.
- 2. <u>Award</u>. The number of Options, the Date of Grant of the Options and the Exercise Price of the Options are set forth at the end of this Award Agreement. The number, if any, of Options that are Incentive Stock Options is also set forth at the end of this Award Agreement. Options that are Incentive Stock Options shall vest and be exercisable at the same times and in the same installments as other Options. One Share shall underlie each Option. THIS AWARD IS SUBJECT TO ALL TERMS, CONDITIONS AND PROVISIONS OF THE PLAN AND THIS AWARD AGREEMENT, INCLUDING, WITHOUT LIMITATION, THE ARBITRATION AND CHOICE OF FORUM PROVISIONS SET FORTH IN PARAGRAPH 18.
- 3. <u>Expiration Date</u>. Notwithstanding anything to the contrary in this Award Agreement, the Options shall expire and no longer be exercisable as of the tenth anniversary of the Date of Grant of the Options (the "*Expiration Date*"), subject to earlier termination as provided in this Award Agreement, or otherwise in accordance with the Plan.

#### 4. Vesting.

- (a) <u>In General</u>. Except as provided below in Paragraph 4(b), you shall become vested in your Options in four equal installments promptly after the first four anniversaries of the Date of Grant set forth at the end of this Award Agreement (each such date, a "*Vesting Date*"). Except as provided in Paragraphs 4(b) and 7(a), if your Employment terminates for any reason prior to a Vesting Date, your rights in respect of all of your Options with respect to which the Vesting Date has not occurred shall terminate, and no such Options shall be exercisable.
- (b) <u>Death</u>. Notwithstanding any other provision of this Award Agreement, if you die prior to an applicable Vesting Date, and provided your rights in respect of your outstanding Options have not previously terminated, as soon as practicable after the date of death and after such documentation as may be requested by the Committee is provided to the Committee, any such outstanding Options that have not vested shall vest.
  - 5. Termination of Options upon Certain Events.
- (a) Except as provided in Paragraphs 4(b) and 7(a), your rights in respect of your outstanding Options for which the Vesting Date has not occurred shall immediately terminate (and no such Options shall be exercisable) if at any time prior to

such Vesting Date your Employment with the Company is terminated for any reason, or you are otherwise no longer actively Employed by the Company.

- (b) Unless the Committee determines otherwise, your rights in respect of all of your outstanding Options (whether or not vested) shall immediately terminate (and no such Options shall be exercisable) if at any time prior to the date you exercise such Options (each such date, an "Exercise Date"):
  - (i) you attempt to have any dispute under this Award Agreement or the Plan resolved in any manner that is not provided for by Paragraph 18; or
  - (ii) any event that constitutes Cause has occurred; or
  - (iii) you in any manner, directly or indirectly, (A) Solicit any Client to transact business with a Competitive Enterprise or to reduce or refrain from doing any business with the Company or (B) interfere with or damage (or attempt to interfere with or damage) any relationship between the Company and any such Client or (C) Solicit any person who is an employee of the Company to resign from the Company or to apply for or accept employment with any Competitive Enterprise; or
  - (iv) you fail to certify to AIG, in accordance with procedures established by the Committee, that you have complied, or the Committee determines that you in fact have failed to comply, with all the terms and conditions of this Award Agreement. By accepting the delivery of Shares (or securities or other property) pursuant to the exercise of Options, you shall be deemed to have represented and certified at such time that you have complied with all the terms and conditions of this Award Agreement.
- 6. <u>Repayment</u>. If, following the delivery of Shares (or securities or other property) pursuant to the exercise of Options, the Committee determines that all terms and conditions of this Award Agreement in respect of such exercise were not satisfied, the Company shall be entitled to receive, and you shall be obligated to pay the Company immediately upon demand therefor, the Fair Market Value of the Shares (determined as of the applicable Exercise Date) or the amount of securities or other property (to the extent that any securities or other property was delivered in lieu of Shares), in each case, underlying the Options exercised, *less* the total exercise price of such Options, but without further reduction of the amount repayable by you for any Shares (or securities or other property) applied to satisfy withholding tax or other obligations in respect of such Shares (or securities or other property).

#### 7. Disability and Retirement.

- (a) Notwithstanding any other provision of this Award Agreement, but subject to Paragraph 7(b), if your Employment with the Company is terminated by reason of Disability or Retirement, the condition set forth in Paragraph 5(a) shall be waived with respect to your then outstanding unvested Options (as a result of which any such then unvested outstanding Options shall vest), but all other conditions of this Award Agreement shall continue to apply.
- (b) Without limiting the application of Paragraph 5(b), your rights in respect of any outstanding Options that become vested solely by reason of Paragraph 7(a) shall terminate, and no such Options shall be exercisable, if, following the termination of your Employment with the Company by reason of Disability or

Retirement and prior to the exercise of such Options, you (i) form, or acquire a 5% or greater equity ownership, voting or profit participation interest in, any Competitive Enterprise or (ii) associate in any capacity (including, but not limited to, association as an officer, employee, partner, director, consultant, agent or advisor) with any Competitive Enterprise.

- 8. <u>Non-transferability</u>. Except as otherwise may be provided by the Committee, the limitations set forth in Section 3.5 of the Plan shall apply. Any assignment or delegation in violation of the provisions of this Paragraph 8 shall be null and void.
  - 9. Exercisability of Vested Options.
- (a) <u>In General</u>. Options that are not vested may not be exercised. Subject to Paragraph 9(b), outstanding vested Options may be exercised in accordance with procedures set forth in Section 2.3.5 of the Plan, including procedures established by the Company.
- (b) <u>Initial Exercise Date</u>. Outstanding vested Options may not be exercised earlier than the first anniversary of the Date of Grant of the Options (the "*Initial Exercise Date*"); *provided* that the condition set forth in this Paragraph 9(b) shall be waived if your employment with the Company is terminated by reason of death, Disability or Retirement as provided in Paragraph 4(b) or 7(a).
- (c) <u>Death, Disability and Retirement</u>. Notwithstanding any other provision of this Award Agreement, if your Employment with the Company is terminated by reason of death, Disability or Retirement and any of your outstanding vested Options remain unexercised, and provided your rights in respect of any such outstanding vested Options have not previously terminated, such outstanding vested Options (including any Options that vest pursuant to Paragraph 4(b) or 7(a)) shall be exercisable, in the case of death, by the representative of your estate in accordance with Paragraph 9(a) beginning on a date that is as soon as practicable after the date of death and after such documentation as may be requested by the Committee is provided to the Committee and, in the case of Disability or Retirement, by you or your legal representative, and shall, unless earlier terminated or cancelled in accordance with the terms of this Agreement, remain exercisable until the Expiration Date and shall thereafter terminate.
- (d) Other Terminations. Subject to Paragraphs 5(b), 7(b) and 9(b), upon the termination of your Employment for any reason (other than death, Disability or Retirement), if any of your outstanding vested Options remain unexercised, and provided your rights in respect of any such outstanding vested Options have not previously terminated, such outstanding vested Options shall be exercisable in accordance with Paragraph 9(a) (but subject to Paragraph 9(b)) and shall, unless earlier terminated or cancelled in accordance with the terms of this Agreement, remain exercisable for 90 days following the termination of your Employment, and shall thereafter terminate.
- 10. <u>Delivery of Shares</u>. Unless otherwise determined by the Committee, or as otherwise provided in this Award Agreement, and except as provided in Paragraphs 11 and 12, upon receipt of payment of the total exercise price of exercised Options in accordance with the procedures set forth in Paragraph 9(a), delivery of Shares underlying such Options shall be effected by the delivery of Certificates representing such Shares. You shall have no right to any dividend, distribution or other right with respect to such Shares if the record date for such dividend, distribution or other right is prior to the date delivery of Certificates

representing such Shares as provided in this Paragraph 10. The Company may postpone such issuance and delivery of Certificates until the Committee is satisfied that you have made such representations and agreements and furnished such information as the Committee may deem necessary pursuant to Section 2.3.5 of the Plan.

- 11. Withholding, Consents and Legends.
- (a) The delivery of Shares underlying your Options is conditioned on your satisfaction of any applicable withholding taxes (in accordance with Section 3.2 of the Plan).
- (b) Your exercise of your Options are conditioned on the receipt to the full satisfaction of the Committee of any required Consent (as defined in Section 3.3 of the Plan) that the Committee may determine to be necessary or advisable (including, without limitation, your consenting to deductions from your wages, or another arrangement satisfactory to the Committee, to reimburse the Company for advances made on your behalf to satisfy withholding and other tax obligations in connection with this Award).
- (c) AIG may affix to Certificates representing Shares issued pursuant to this Award Agreement any legend that the Committee determines to be necessary or advisable (including to reflect any restrictions to which you may be subject under a separate agreement with AIG). AIG may advise the transfer agent to place a stop transfer order against any legended Shares.
- 12. <u>Right of Offset</u>. The Company shall have the right to offset against the obligation to deliver Shares (or securities or other property) under this Award Agreement any outstanding amounts (including, without limitation, travel and entertainment or advance account balances, loans, or amounts repayable to the Company pursuant to tax equalization, housing, automobile or other employee programs) you then owe to the Company and any amounts the Committee otherwise deems appropriate.
- 13. No Rights to Continued Employment. Nothing in this Award Agreement or the Plan shall be construed as giving you any right to continued Employment by the Company or affect any right that the Company may have to terminate or alter the terms and conditions of your Employment.
- 14. <u>Successors and Assigns of AIG</u>. The terms and conditions of this Award Agreement shall be binding upon, and shall inure to the benefit of, AIG and its successor entities (as defined in Section 3.6 of the Plan).
- 15. <u>Committee Discretion</u>. The Committee shall have full discretion with respect to any actions to be taken or determinations to be made in connection with this Award Agreement, and its determinations shall be final, binding and conclusive.
- 16. <u>Amendment</u>. The Committee reserves the right at any time to amend the terms and conditions set forth in this Award Agreement, and the Board may amend the Plan in any respect; *provided* that, notwithstanding the foregoing and Sections 1.3.1(a), 1.3.1(b) and 3.1 of the Plan, no such amendment shall materially adversely affect your rights and obligations under this Award Agreement without your consent. Any amendment of this Award Agreement shall be in writing signed by an authorized member of the Committee or a person or persons designated by the Committee.

- 17. Adjustment. Your Options shall be subject to amendment and adjustment in accordance with Section 1.6.4 of the Plan.
- 18. Arbitration; Choice of Forum.
- (a) Any dispute, controversy or claim between the Company and you, arising out of or relating to or concerning the Plan or this Award Agreement, shall be finally settled by arbitration in New York City before, and in accordance with the rules then obtaining of, the New York Stock Exchange, Inc. (the "NYSE") or, if the NYSE declines to arbitrate the matter (or if the matter otherwise is not arbitrable by it), the American Arbitration Association (the "AAA") in accordance with the commercial arbitration rules of the AAA. Prior to arbitration, all claims maintained by you must first be submitted to the Committee in accordance with claims procedures determined by the Committee. This paragraph is subject to the provisions of Paragraphs 18(b) and (c) below.
- (b) THE COMPANY AND YOU HEREBY IRREVOCABLY SUBMIT TO THE EXCLUSIVE JURISDICTION OF ANY STATE OR FEDERAL COURT LOCATED IN THE CITY OF NEW YORK OVER ANY SUIT, ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO OR CONCERNING THE PLAN OR THIS AWARD AGREEMENT THAT IS NOT OTHERWISE ARBITRATED OR RESOLVED ACCORDING TO PARAGRAPH 18(a) OF THIS AWARD AGREEMENT. This includes any suit, action or proceeding to compel arbitration or to enforce an arbitration award. The Company and you acknowledge that the forum designated by this Paragraph 18(b) has a reasonable relation to the Plan, this Award Agreement, and to your relationship with the Company. Notwithstanding the foregoing, nothing herein shall preclude the Company from bringing any action, suit or proceeding in any other court for the purpose of enforcing the provisions of this Paragraph 18.
- (c) The agreement by you and the Company as to forum is independent of the law that may be applied in the action, suit or proceeding and you and the Company agree to such forum even if the forum may under applicable law choose to apply non-forum law. You and the Company hereby waive, to the fullest extent permitted by applicable law, any objection which you or the Company now or hereafter may have to personal jurisdiction or to the laying of venue of any such suit, action or proceeding in any court referred to in Paragraph 18(b). You and the Company undertake not to commence any action, suit or proceeding arising out of or relating to or concerning this Award Agreement in any forum other than a forum described in this Paragraph 18. You and (subject to the last sentence of Paragraph 18(b)) the Company agree that, to the fullest extent permitted by applicable law, a final and non-appealable judgment in any such suit, action or proceeding in any such court shall be conclusive and binding upon you and the Company.
- (d) You irrevocably appoint the Secretary of AIG as your agent for service of process in connection with any action, suit or proceeding arising out of or relating to or concerning this Award Agreement which is not arbitrated pursuant to the provisions of Paragraph 18(a), who shall promptly advise you of any such service of process.
- (e) You hereby agree to keep confidential the existence of, and any information concerning, a dispute described in this Paragraph 18, except that you may disclose information concerning such dispute to the arbitrator or court that is considering such dispute or to your legal counsel (*provided* that such counsel agrees not to disclose any such information other than as necessary to the prosecution or defense of the dispute).

(f) You recognize and agree that prior to the grant of this Award you have no right to any benefits hereunder. Accordingly, in consideration of the receipt of this Award, you expressly waive any right to contest the amount of this Award, terms of this Award Agreement, any determination, action or omission hereunder or under the Plan by the Committee, AIG or the Board, or any amendment to the Plan or this Award Agreement (other than an amendment to which your consent is expressly required by Paragraph 16) and you expressly waive any claim related in any way to the Award including any claim based on any promissory estoppel or other theory in connection with this Award and your Employment with the Company.

## 19. <u>Governing Law.</u> THIS AWARD SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK, WITHOUT REGARD TO PRINCIPLES OF CONFLICTS OF LAWS.

20. <u>Headings</u>. The headings in this Award Agreement are for the purpose of convenience only and are not intended to define or limit the construction of the provisions hereof.

Date of Grant.				
				AMERICAN INTERNATIONAL GROUP, INC.
				Ву:
				Ву:
Recipient:				
Date of Grant:	-			
Exercise Price:				
Number of Options:	·			
Of the above Options,  Number of Options:	_	difficer are fricer		15.
Receipt Acknowledged:				
Address:		Ctoolst		
		Street		
	City,	State	Zip Code	
	Social S	Security No./Loc	al I.D. No.	
				-7-

IN WITNESS WHEREOF, AMERICAN INTERNATIONAL GROUP, INC. has caused this Award Agreement to be duly executed and delivered as of the

#### **Glossary of Terms**

Solely for purposes of this award of RSUs, the following terms shall have the meanings set forth below. Capitalized terms not defined in this Glossary of Terms shall have the meanings as used or defined in the applicable Award Agreement or the Plan.

"Cause" means (i) your conviction, whether following trial or by plea of guilty or nolo contendere (or similar plea), in a criminal proceeding (A) on a misdemeanor charge involving fraud, false statements or misleading omissions, wrongful taking, embezzlement, bribery, forgery, counterfeiting or extortion, or (B) on a felony charge or (C) on an equivalent charge to those in clauses (A) and (B) in jurisdictions which do not use those designations; (ii) your engaging in any conduct which constitutes an employment disqualification under applicable law (including statutory disqualification as defined under the Exchange Act); (iii) your failure to perform your duties to the Company; (iv) your violation of any securities or commodities laws, any rules or regulations issued pursuant to such laws, or the rules and regulations of any securities or commodities exchange or association of which AIG or any of its subsidiaries or affiliates is a member; (v) your violation of any Company policy concerning hedging or confidential or proprietary information, or your material violation of any other Company policy as in effect from time to time; (vi) your engaging in any act or making any statement which impairs, impugns, denigrates, disparages or negatively reflects upon the name, reputation or business interests of the Company; or (vii) your engaging in any conduct detrimental to the Company. The determination as to whether "Cause" has occurred shall be made by the Committee in its sole discretion. The Committee shall also have the authority in its sole discretion to waive the consequences under the Plan or any Award Agreement of the existence or occurrence of any of the events, acts or omissions constituting "Cause."

"Client" means any client or prospective client of the Company to whom you provided services, or for whom you transacted business, or whose identity became known to you in connection with your relationship with or Employment by the Company.

"Competitive Enterprise" means a business enterprise that (i) engages in any activity, or (ii) owns or controls a significant interest in any entity that engages in any activity, that, in either case, competes anywhere with any activity in which the Company is engaged. The activities covered by the previous sentence include, without limitation, all insurance and re-insurance and re-insurance-related activities, asset management, financial product activities (including, without limitation, derivative activities) and financial services in the United States and abroad.

"Disability" means "permanent disability" as defined in the American International Group, Inc. Group Long — Term Insurance Policy as in effect on the Date of Grant.

"Retirement" means "normal retirement" as defined in the American International Group, Inc. Retirement Plan as in effect on the Date of Grant.

"Solicit" means any direct or indirect communication of any kind whatsoever, regardless of by whom initiated, inviting, advising, encouraging or requesting any person or entity, in any manner, to take or refrain from taking any action.

# AMERICAN INTERNATIONAL GROUP, INC. PARTNERS PLAN PERFORMANCE RSU AWARD AGREEMENT

This award agreement (this "Award Agreement") sets forth the terms and conditions of an award (this "Award") of performance-based restricted stock units ("Performance RSUs") awarded to you pursuant to the American International Group, Inc. Partners Plan (the "Partners Plan") and issued to you under the American International Group, Inc. Amended and Restated 2007 Stock Incentive Plan (the "SIP").

1. <u>The SIP</u>. This Award is issued under the SIP, the terms of which are incorporated in this Award Agreement. Capitalized terms used in this Award Agreement that are not defined in this Award Agreement, or in the attached Glossary of Terms, have the meanings as used or defined in the SIP.

#### 2. Award.

- (a) *Award of Performance RSUs*. The number of Performance RSUs initially subject to this Award, the Date of Grant and the Performance Period are set forth at the end of this Award Agreement. At the end of the Performance Period, the Committee will determine the number of Performance RSUs earned under this Award in accordance with the terms of the Partners Plan and will advise you of the number (such earned Performance RSUs, the "*Earned RSUs*"). The number of Earned RSUs may range from 0% to 150% of the Performance RSUs initially subject to this Award.
- (b) Status of Earned RSUs. Each Earned RSU constitutes an unfunded and unsecured promise of AIG to deliver (or cause to be delivered) to you, subject to the terms of this Award Agreement, one share of Common Stock (the "Share" or the "Shares" as the context requires) (or securities or other property equal to the Fair Market Value thereof) on the Scheduled Vesting Date as provided herein. Until such delivery, you have only the rights of a general unsecured creditor, and no rights as a shareholder, of AIG. THIS AWARD IS SUBJECT TO ALL TERMS, CONDITIONS AND PROVISIONS OF THE PARTNERS PLAN, THE SIP AND THIS AWARD AGREEMENT, INCLUDING, WITHOUT LIMITATION, THE ARBITRATION AND CHOICE OF FORUM PROVISIONS SET FORTH IN PARAGRAPH 15.

#### 3. Vesting and Delivery.

- (a) *Vesting*. Except as provided in this Paragraph 3 and in Paragraphs 4 and 6, you shall become vested in the Earned RSUs, and the Shares underlying the Earned RSUs shall be delivered, in two equal installments promptly after the fourth and sixth anniversaries of the first day of the Performance Period for this Award (each, a "*Scheduled Vesting Date*" for this Award). Except as provided in Paragraph 6, if your Employment terminates for any reason prior to a Scheduled Vesting Date, your rights in respect of all of your unvested Performance RSUs shall be forfeited and terminate, and no Shares (or securities or other property) shall be delivered in respect of such Performance RSUs.
- (b) *Delivery*. Except as provided in this Paragraph 3 and in Paragraphs 4, 6, 8 and 9, the Shares underlying the vested Earned RSUs shall be delivered on the respective Scheduled Vesting Date. The Company may, at its option, deliver securities or other property in lieu of all or any portion of the Shares otherwise deliverable on a Scheduled Vesting Date. Such payment shall equal the product of the

number of Shares to be delivered on the Scheduled Vesting Date and the Fair Market Value of one Share of Common Stock on the Scheduled Vesting Date. You shall be deemed the beneficial owner of the Shares at the close of business on the Scheduled Vesting Date and shall be entitled to any dividend or distribution that has not already been made with respect to such Shares if the record date for such dividend or distribution is on or after the close of business on the Scheduled Vesting Date. Notwithstanding the foregoing, if a Scheduled Vesting Date occurs at a time when the tax deduction with respect to your delivery of Shares (or securities or other property) would be limited or eliminated by Section 162(m) of the Code, then, delivery of the Shares (or securities or other property) automatically shall be deferred until the earliest date that the Company reasonably anticipates that the deduction or payment will not be limited or eliminated.

- (c) **Death**. Notwithstanding any other provision of this Award Agreement, if you die:
- (i) during the Performance Period, and provided your rights in respect of your Performance RSUs have not yet terminated, (A) the Earned RSUs in respected of your Performance RSUs will be determined on a pro-rata basis (based upon the number of whole or partial months you were employed during the Performance Period relative to 24) and (B) when the Earned RSUs have been determined in accordance with Paragraphs 2(c)(i)(A) and 2(b), the Shares (or securities or other property in lieu of all or any part thereof) corresponding to such Earned RSUs shall be delivered to the representative of your estate as soon as practicable after the date of such determination and after such documentation as may be requested by the Committee is provided to the Committee; or
- (ii) at any time prior to a Scheduled Vesting Date but after completion of the Performance Period, and provided your rights in respect of your Earned RSUs have not yet terminated, the Shares (or securities or other property in lieu of all or any part thereof) corresponding to your outstanding Earned RSUs shall be delivered to the representative of your estate as soon as practicable after the date of death and after such documentation as may be requested by the Committee is provided to the Committee.
- 4. Termination of Performance RSUs and Earned RSUs; Non-Delivery of Shares.
- (a) *Termination on Separation from Service*. Except as provided in Paragraphs 3(c) and 6, if your Employment with the Company terminates for any reason or you are otherwise no longer actively employed by the Company:
  - (i) during the Performance Period, your rights in respect of your outstanding Performance RSUs shall immediately terminate, there will be no Earned RSUs with respect thereto and no Shares (or securities or other property) shall be delivered in respect thereof; or
  - (ii) at any time prior to a Scheduled Vesting Date but after the completion of the Performance Period, your rights in respect of your outstanding unvested Earned RSUs shall immediately

terminate, and no Shares (or securities or other property) shall be delivered in respect of such unvested Earned RSUs.

- (b) *Termination on Other Events*. Unless the Committee determines otherwise, and except as further provided in Paragraph 6, your rights in respect of all of your Performance RSUs or Earned RSUs (whether or not vested) shall immediately terminate, and no Shares (or securities or other property) shall be delivered in respect of Performance RSUs or Earned RSUs, if at any time prior to the Scheduled Vesting Date:
  - (i) you attempt to have any dispute under this Award Agreement, the Partners Plan or the SIP resolved in any manner that is not provided for by Paragraph 15; or
  - (ii) any event that constitutes Cause has occurred; or
  - (iii) you in any manner, directly or indirectly, (A) Solicit any Client to transact business with a Competitive Enterprise or to reduce or refrain from doing any business with the Company or (B) interfere with or damage (or attempt to interfere with or damage) any relationship between the Company and any such Client or (C) Solicit any person who is an employee of the Company to resign from the Company or to apply for or accept employment with any Competitive Enterprise; or
  - (iv) you fail to certify to AIG, in accordance with procedures established by the Committee, with respect to a Scheduled Vesting Date that you have complied, or the Committee determines that you have failed as of a Scheduled Vesting Date to comply, with all of the terms and conditions of this Award Agreement. By accepting the delivery of Shares (or securities or other property) under this Award Agreement, you shall be deemed to have represented and certified at such time that you have complied with all the terms and conditions of this Award Agreement.
- (c) *Termination on Failure to Certify.* Unless the Committee determines otherwise, if a Scheduled Vesting Date in respect of any of your outstanding Earned RSUs occurs, and Shares (or securities or other property) with respect to such outstanding Earned RSUs would be deliverable under the terms and conditions of this Award Agreement, except that you have not complied with the conditions or your obligations under Paragraph 4(b)(iv), all of your rights with respect to your outstanding Earned RSUs shall terminate no later than the Scheduled Vesting Date for such Shares.
- 5. <u>Repayment</u>. If, following the delivery of Shares (or securities or other property), the Committee determines that all terms and conditions of this Award Agreement in respect of such delivery were not satisfied, the Company shall be entitled to receive, and you shall be obligated to pay the Company immediately upon demand therefor, the Fair Market Value of the Shares (determined as of the Scheduled Vesting Date) and the amount of securities or other property (to the extent that securities or other property was delivered in lieu of Shares) delivered with respect to the Scheduled Vesting Date, without reduction for any Shares (or securities or other property) applied to satisfy withholding tax or other obligations in respect of such Shares (or securities or other property).

#### 6. Disability and Retirement.

- (a) *During the Performance Period*. Notwithstanding any other provision of this Award Agreement, but subject to Paragraph 6(c), if your employment with the Company is terminated by reason of Disability or Retirement during the Performance Period, (i) the Earned RSUs in respected of your Performance RSUs will be determined on a pro-rata basis (based upon the number of whole or partial months you were employed during the Performance Period relative to 24) and (ii) when the Earned RSUs have been determined in accordance with Paragraph 6(a)(i) and Paragraph 2(b), the condition set forth in Paragraph 4(a) shall be waived with respect to any such Earned RSUs (as a result of which any such Earned RSUs shall vest and shares corresponding to the Earned RSUs shall be delivered to you as soon as practicable after the date of termination and after such documentation as may be requested by the Committee is provided to the Committee), but all other conditions of this Award Agreement shall continue to apply.
- (b) *After the End of the Performance Period.* Notwithstanding any other provision of this Award Agreement, but subject to Paragraph 6(c), if your employment with the Company is terminated by reason of Disability or Retirement after the completion of the Performance Period, the condition set forth in Paragraph 4(a) shall be waived with respect to your then outstanding unvested Earned RSUs (as a result of which any such then unvested outstanding Earned RSUs shall vest and shares corresponding to the Earned RSUs shall be delivered to you as soon as practicable after the date of termination and after such documentation as may be requested by the Committee is provided to the Committee), but all other conditions of this Award Agreement shall continue to apply.
- (c) *Termination of Rights Following Disability or Retirement.* Without limiting the application of Paragraph 4(b) or Paragraph 4(c), your rights in respect of any outstanding Earned RSUs that become vested solely by reason of Paragraph 6(a) shall terminate, and no Shares (or securities or other property) shall be delivered in respect of such outstanding Earned RSUs if, following the termination of your Employment with the Company by reason of Disability or Retirement and prior to the Scheduled Vesting Date, you (i) form, or acquire a 5% or greater equity ownership, voting or profit participation interest in, any Competitive Enterprise or (ii) associate in any capacity (including, but not limited to, association as an officer, employee, partner, director, consultant, agent or advisor) with any Competitive Enterprise.
- 7. <u>Non-transferability</u>. Except as otherwise may be provided by the Committee, the limitations set forth in Section 8L of the Partners Plan shall apply. Any assignment or delegation in violation of the provisions of this Paragraph 7 shall be null and void.
  - 8. Withholding, Consents and Legends.
    - (a) The delivery of Shares is conditioned on your satisfaction of any applicable withholding taxes (in accordance with Section 3.2 of the SIP).
- (b) Your rights in respect of your Performance RSUs and Earned RSUs are conditioned on the receipt to the full satisfaction of the Committee of any required consents (as defined in Section 8N of the Partners Plan) that the Committee may determine to be necessary or advisable (including, without limitation, your consenting to deductions from your wages, or another arrangement satisfactory to the

Committee, to reimburse the Company for advances made on your behalf to satisfy withholding and other tax obligations in connection with this Award).

- (c) AIG may affix to Certificates representing Shares issued pursuant to this Award Agreement any legend that the Committee determines to be necessary or advisable (including to reflect any restrictions to which you may be subject under a separate agreement with AIG). AIG may advise the transfer agent to place a stop transfer order against any legended Shares.
- 9. <u>Right of Offset</u>. The Company shall have the right to offset against the obligation to deliver Shares (or securities or other property) under this Award Agreement any outstanding amounts (including, without limitation, travel and entertainment or advance account balances, loans, or amounts repayable to the Company pursuant to tax equalization, housing, automobile or other employee programs) you then owe to the Company and any amounts the Committee otherwise deems appropriate.
- 10. No Rights to Continued Employment. Nothing in this Award Agreement, the Partners Plan or the SIP shall be construed as giving you any right to continued Employment by the Company or affect any right that the Company may have to terminate or alter the terms and conditions of your Employment or to be included in any future plans of similar nature.
- 11. <u>Successors and Assigns of AIG</u>. The terms and conditions of this Award Agreement shall be binding upon, and shall inure to the benefit of, AIG and its successor entities (as defined in Section 3.6 of the SIP).
- 12. <u>Committee Discretion</u>. The Committee shall have full discretion with respect to any actions to be taken or determinations to be made in connection with this Award Agreement, and its determinations shall be final, binding and conclusive.
- 13. <u>Amendment</u>. The Committee reserves the right at any time and in any manner to amend the terms and conditions set forth in this Award Agreement and the Partners Plan in accordance with Sections 1.3.1 of the SIP, including in a manner adverse to your rights.
  - 14. Adjustment. The Performance RSUs shall be subject to amendment and adjustment in accordance with Section 1.6.4 of the SIP.
  - 15. Arbitration; Choice of Forum.
- (a) Any dispute, controversy or claim between the Company and you, arising out of or relating to or concerning the SIP or this Award Agreement, shall be finally settled by arbitration in New York City before, and in accordance with the rules then obtaining of, the New York Stock Exchange, Inc. (the "NYSE") or, if the NYSE declines to arbitrate the matter (or if the matter otherwise is not arbitrable by it), the American Arbitration Association (the "AAA") in accordance with the commercial arbitration rules of the AAA. Prior to arbitration, all claims maintained by you must first be submitted to the Committee in accordance with claims procedures determined by the Committee. This paragraph is subject to the provisions of Paragraphs 15(b) and (c) below.
- (b) THE COMPANY AND YOU HEREBY IRREVOCABLY SUBMIT TO THE EXCLUSIVE JURISDICTION OF ANY STATE OR FEDERAL COURT LOCATED

IN THE CITY OF NEW YORK OVER ANY SUIT, ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO OR CONCERNING THE SIP OR THIS AWARD AGREEMENT THAT IS NOT OTHERWISE ARBITRATED OR RESOLVED ACCORDING TO PARAGRAPH 15(a) OF

**THIS AWARD AGREEMENT.** This includes any suit, action or proceeding to compel arbitration or to enforce an arbitration award. The Company and you acknowledge that the forum designated by this Paragraph 15(b) has a reasonable relation to the SIP, this Award Agreement, and to your relationship with the Company. Notwithstanding the foregoing, nothing herein shall preclude the Company from bringing any action, suit or proceeding in any other court for the purpose of enforcing the provisions of this Paragraph 15.

- (c) The agreement by you and the Company as to forum is independent of the law that may be applied in the action, suit or proceeding, and you and the Company agree to such forum even if the forum may under applicable law choose to apply non-forum law. You and the Company hereby waive, to the fullest extent permitted by applicable law, any objection which you or the Company now or hereafter may have to personal jurisdiction or to the laying of venue of any such suit, action or proceeding in any court referred to in Paragraph 15(b). You and the Company undertake not to commence any action, suit or proceeding arising out of or relating to or concerning this Award Agreement in any forum other than a forum described in this Paragraph 15. You and (subject to the last sentence of Paragraph 15(b)) the Company agree that, to the fullest extent permitted by applicable law, a final and non-appealable judgment in any such suit, action or proceeding in any such court shall be conclusive and binding upon you and the Company.
- (d) You irrevocably appoint the Secretary of AIG as your agent for service of process in connection with any action, suit or proceeding arising out of or relating to or concerning the Partners Plan or the Award Agreement which is not arbitrated pursuant to the provisions of Paragraph 15(a), who shall promptly advise you of any such service of process.
- (e) You hereby agree to keep confidential the existence of, and any information concerning any grant made under the Partners Plan and any dispute, controversy or claim relating to the Partners Plan or this Award Agreement, except that you may disclose information concerning such dispute or claim to the arbitrator or court that is considering such dispute or to your legal counsel (provided that such counsel agrees not to disclose any such information other than as necessary to the prosecution or defense of the dispute).
- (f) You recognize and agree that prior to the grant of this Award you have no right to any benefits hereunder. Accordingly, in consideration of the receipt of this Award, you expressly waive any right to contest the amount of this Award, terms of this Award Agreement and the Partners Plan, any determination, action or omission hereunder or under the SIP by the Committee or AIG of the Board, or any amendment to the Partners Plan or this Award Agreement and you expressly waive any claim related in any way to the Award including any claim based on any promissory estoppel or other theory in connection with this Award and your Employment with the Company.

16. <u>Governing Law</u>. THIS AWARD SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK, WITHOUT REGARD TO PRINCIPLES OF CONFLICTS OF LAWS.

17. <u>Headings</u> . The headings in this Award Approvisions hereof.	Agreement are for the purpose of convenience only and are not intended	d to define or limit the construction of the
	-7-	

IN WITNESS WHEREOF, AMERICAN INTERNATIONAL	GROUP, INC. has caused this Award Agreement to be duly executed and delivered as of the
Date of Grant.	

By: Nan				
Title				
Ву:				
Nan				
Title	5:			

Recipient:

Number of Performance RSUs:
Max. Earned RSUs
Target Earned RSUs
Threshold Earned RSUs
Min. Earned RSUs
0
Performance Period:
Date of Grant:
Scheduled Vesting Date:

#### Glossary of Terms

Solely for purposes of this award of Performance RSUs, the following terms shall have the meanings set forth below. Capitalized terms not defined in this Glossary of Terms shall have the meanings as used or defined in the applicable Award Agreement or the SIP.

"Cause" means (i) your conviction, whether following trial or by plea of guilty or nolo contendere (or similar plea), in a criminal proceeding (A) on a misdemeanor charge involving fraud, false statements or misleading omissions, wrongful taking, embezzlement, bribery, forgery, counterfeiting or extortion, or (B) on a felony charge or (C) on an equivalent charge to those in clauses (A) and (B) in jurisdictions which do not use those designations; (ii) your engaging in any conduct which constitutes an employment disqualification under applicable law (including statutory disqualification as defined under the Exchange Act); (iii) your failure to perform your duties to the Company; (iv) your violation of any securities or commodities laws, any rules or regulations issued pursuant to such laws, or the rules and regulations of any securities or commodities exchange or association of which AIG or any of its subsidiaries or affiliates is a member; (v) your violation of any Company policy concerning hedging or confidential or proprietary information, or your material violation of any other Company policy as in effect from time to time; (vi) your engaging in any act or making any statement which impairs, impugns, denigrates, disparages or negatively reflects upon the name, reputation or business interests of the Company; or (vii) your engaging in any conduct detrimental to the Company. The determination as to whether "Cause" has occurred shall be made by the Committee in its sole discretion. The Committee shall also have the authority in its sole discretion to waive the consequences under the Partners Plan, SIP or any Award Agreement of the existence or occurrence of any of the events, acts or omissions constituting "Cause."

"*Client*" means any client or prospective client of the Company to whom you provided services, or for whom you transacted business, or whose identity became known to you in connection with your relationship with or Employment by the Company.

"Competitive Enterprise" means a business enterprise that (i) engages in any activity, or (ii) owns or controls a significant interest in any entity that engages in any activity, that, in either case, competes anywhere with any activity in which the Company is engaged. The activities covered by the previous sentence include, without limitation, all insurance and re-insurance and re-insurance-related activities, asset management, financial product activities (including, without limitation, derivative activities) and financial services in the United States and abroad.

"Disability" means "permanent disability" as defined in the American International Group, Inc. Group Long — Term Insurance Policy as in effect on the Date of Grant.

"Retirement" means "normal retirement" as defined in the American International Group, Inc. Retirement Plan as in effect on the Date of Grant.

"Solicit" means any direct or indirect communication of any kind whatsoever, regardless of by whom initiated, inviting, advising, encouraging or requesting any person or entity, in any manner, to take or refrain from taking any action.

# AMERICAN INTERNATIONAL GROUP, INC. AMENDED AND RESTATED 2007 STOCK INCENTIVE PLAN RSU AWARD AGREEMENT

This award agreement (this "Award Agreement") sets forth the terms and conditions of an award (this "Award") of restricted stock units ("RSUs") granted to you under the American International Group, Inc. Amended and Restated 2007 Stock Incentive Plan (the "Plan").

- 1. <u>The Plan</u>. This Award is made pursuant to the Plan, the terms of which are incorporated in this Award Agreement. Capitalized terms used in this Award Agreement that are not defined in this Award Agreement, or in the attached Glossary of Terms, have the meanings as used or defined in the Plan.
- 2. <u>Award</u>. The number of RSUs subject to this Award is set forth at the end of this Award Agreement. Each RSU constitutes an unfunded and unsecured promise of AIG to deliver (or cause to be delivered) to you, subject to the terms of this Award Agreement, one share of Common Stock (the "Share" or the "Shares" as the context requires) (or securities or other property equal to the Fair Market Value thereof) on the Scheduled Vesting Date as provided herein. Until such delivery, you have only the rights of a general unsecured creditor, and no rights as a shareholder, of AIG. THIS AWARD IS SUBJECT TO ALL TERMS, CONDITIONS AND PROVISIONS OF THE PLAN AND THIS AWARD AGREEMENT, INCLUDING, WITHOUT LIMITATION, THE ARBITRATION AND CHOICE OF FORUM PROVISIONS SET FORTH IN PARAGRAPH 15.
  - 3. Vesting and Delivery.
- (a) <u>Vesting</u>. Except as provided in this Paragraph 3 and in Paragraphs 4 and 6, you shall become vested in the RSUs, and the Shares underlying the RSUs shall be delivered, on the fourth anniversary of the Date of Grant specified at the end of this Award Agreement (the "*Scheduled Vesting Date*"). Except as provided in Paragraph 6, if your Employment terminates for any reason prior to the Scheduled Vesting Date, your rights in respect of all of your RSUs shall terminate, and no Shares (or securities or other property) shall be delivered in respect of such RSUs.
- (b) <u>Delivery.</u> Except as provided in this Paragraph 3 and in Paragraphs 4, 6, 8 and 9, the Shares underlying the vested RSUs shall be delivered on the Scheduled Vesting Date. The Company may, at its option, deliver securities or other property in lieu of all or any portion of the Shares otherwise deliverable on the Scheduled Vesting Date. Such payment shall equal the product of the number of Shares to be delivered on the Scheduled Vesting Date and the Fair Market Value of one Share of Common Stock on the Scheduled Vesting Date. You shall be deemed the beneficial owner of the Shares at the close of business on the Scheduled Vesting Date and shall be entitled to any dividend or distribution that has not already been made with respect to such Shares if the record date for such dividend or distribution is on or after the close of business on the Scheduled Vesting Date. Notwithstanding the foregoing, if the Scheduled Vesting Date occurs at a time when you are considered by AIG to be one of its "covered employees" within the meaning of Section 162(m) of the Code, then, unless the Committee determines otherwise, delivery of the Shares (or securities or other property) automatically shall be deferred until after you have ceased to be such a covered employee.

(c) <u>Death</u>. Notwithstanding any other provision of this Award Agreement, if you die prior to the Scheduled Vesting Date, and provided your rights in respect of your RSUs have not previously terminated, the Shares (or securities or other property in lieu of all or any part thereof) corresponding to your outstanding RSUs shall be delivered to the representative of your estate as soon as practicable after the date of death and after such documentation as may be requested by the Committee is provided to the Committee.

#### 4. Termination of RSUs and Non-Delivery of Shares.

- (a) Except as provided in Paragraphs 3(c) and 6, your rights in respect of your outstanding RSUs shall immediately terminate, and no Shares (or securities or other property) shall be delivered in respect of such unvested RSUs, if at any time prior to the Scheduled Vesting Date your Employment with the Company terminates for any reason, or you are otherwise no longer actively Employed by the Company.
- (b) Unless the Committee determines otherwise, and except as further provided in Paragraph 6, your rights in respect of all of your RSUs (whether or not vested) shall immediately terminate, and no Shares (or securities or other property) shall be delivered in respect of such RSUs, if at any time prior to the Scheduled Vesting Date:
  - (i) you attempt to have any dispute under this Award Agreement or the Plan resolved in any manner that is not provided for by Paragraph 15; or
  - (ii) any event that constitutes Cause has occurred; or
  - (iii) you in any manner, directly or indirectly, (A) Solicit any Client to transact business with a Competitive Enterprise or to reduce or refrain from doing any business with the Company or (B) interfere with or damage (or attempt to interfere with or damage) any relationship between the Company and any such Client or (C) Solicit any person who is an employee of the Company to resign from the Company or to apply for or accept employment with any Competitive Enterprise; or
  - (iv) you fail to certify to AIG, in accordance with procedures established by the Committee, with respect to the Scheduled Vesting Date that you have complied, or the Committee determines that you have failed as of the Scheduled Vesting Date to comply, with all of the terms and conditions of this Award Agreement. By accepting the delivery of Shares (or securities or other property) under this Award Agreement, you shall be deemed to have represented and certified at such time that you have complied with all the terms and conditions of this Award Agreement.
- (c) Unless the Committee determines otherwise, if the Scheduled Vesting Date in respect of any of your outstanding RSUs occurs, and Shares (or securities or other property) with respect to such outstanding RSUs would be deliverable under the terms and conditions of this Award Agreement, except that you have not complied with the conditions or your obligations under Paragraph 4(b)(iv), all of your rights with respect to your outstanding RSUs shall terminate no later than the Scheduled Vesting Date for such Shares.

5. <u>Repayment</u>. If, following the delivery of Shares (or securities or other property), the Committee determines that all terms and conditions of this Award Agreement in respect of such delivery were not satisfied, the Company shall be entitled to receive, and you shall be obligated to pay the Company immediately upon demand therefor, the Fair Market Value of the Shares (determined as of the Scheduled Vesting Date) and the amount of securities or other property (to the extent that any securities or other property was delivered in lieu of Shares) delivered with respect to the Scheduled Vesting Date, without reduction for any Shares (or securities or other property) applied to satisfy withholding tax or other obligations in respect of such Shares (or securities or other property).

#### 6. Disability and Retirement.

- (a) Notwithstanding any other provision of this Award Agreement, but subject to Paragraph 6(b), if your Employment with the Company is terminated by reason of Disability or Retirement, the condition set forth in Paragraph 4(a) shall be waived with respect to your then outstanding unvested RSUs (as a result of which any such then unvested outstanding RSUs shall vest and shares corresponding to the RSUs shall be delivered to you as soon as practicable after the date of termination and after such documentation as may be requested by the Committee is provided to the Committee), but all other conditions of this Award Agreement shall continue to apply.
- (b) Without limiting the application of Paragraph 4(b) or Paragraph 4(c), your rights in respect of any outstanding RSUs that become vested solely by reason of Paragraph 6(a) shall terminate, and no Shares (or securities or other property) shall be delivered in respect of such outstanding RSUs if, following the termination of your Employment with the Company by reason of Disability or Retirement and prior to the Scheduled Vesting Date, you (i) form, or acquire a 5% or greater equity ownership, voting or profit participation interest in, any Competitive Enterprise or (ii) associate in any capacity (including, but not limited to, association as an officer, employee, partner, director, consultant, agent or advisor) with any Competitive Enterprise.
- 7. <u>Non-transferability</u>. Except as otherwise may be provided by the Committee, the limitations set forth in Section 3.5 of the Plan shall apply. Any assignment or delegation in violation of the provisions of this Paragraph 7 shall be null and void.
  - 8. Withholding, Consents and Legends.
  - (a) The delivery of Shares is conditioned on your satisfaction of any applicable withholding taxes (in accordance with Section 3.2 of the Plan).
- (b) Your rights in respect of your RSUs are conditioned on the receipt to the full satisfaction of the Committee of any required Consent (as defined in Section 3.3 of the Plan) that the Committee may determine to be necessary or advisable (including, without limitation, your consenting to deductions from your wages, or another arrangement satisfactory to the Committee, to reimburse the Company for advances made on your behalf to satisfy withholding and other tax obligations in connection with this Award).
- (c) AIG may affix to Certificates representing Shares issued pursuant to this Award Agreement any legend that the Committee determines to be necessary or advisable (including to reflect any restrictions to which you may be subject under a separate agreement with AIG). AIG may advise the transfer agent to place a stop transfer order against any legended Shares.

- 9. <u>Right of Offset</u>. The Company shall have the right to offset against the obligation to deliver Shares (or securities or other property) under this Award Agreement any outstanding amounts (including, without limitation, travel and entertainment or advance account balances, loans, or amounts repayable to the Company pursuant to tax equalization, housing, automobile or other employee programs) you then owe to the Company and any amounts the Committee otherwise deems appropriate.
- 10. No Rights to Continued Employment. Nothing in this Award Agreement or the Plan shall be construed as giving you any right to continued Employment by the Company or affect any right that the Company may have to terminate or alter the terms and conditions of your Employment.
- 11. <u>Successors and Assigns of AIG</u>. The terms and conditions of this Award Agreement shall be binding upon, and shall inure to the benefit of, AIG and its successor entities (as defined in Section 3.6 of the Plan).
- 12. <u>Committee Discretion</u>. The Committee shall have full discretion with respect to any actions to be taken or determinations to be made in connection with this Award Agreement, and its determinations shall be final, binding and conclusive.
- 13. <u>Amendment</u>. The Committee reserves the right at any time to amend the terms and conditions set forth in this Award Agreement, and the Board may amend the Plan in any respect; provided, that, notwithstanding the foregoing and Sections 1.3.1(a), 1.3.1(b) and 3.1 of the Plan, no such amendment shall materially adversely affect your rights and obligations under this Award Agreement without your consent. Any amendment of this Award Agreement shall be in writing signed by an authorized member of the Committee or a person or persons designated by the Committee.
  - 14. Adjustment. The RSUs shall be subject to amendment and adjustment in accordance with Section 1.6.4 of the Plan.
  - 15. Arbitration; Choice of Forum.
- (a) Any dispute, controversy or claim between the Company and you, arising out of or relating to or concerning the Plan or this Award Agreement, shall be finally settled by arbitration in New York City before, and in accordance with the rules then obtaining of, the New York Stock Exchange, Inc. (the "NYSE") or, if the NYSE declines to arbitrate the matter (or if the matter otherwise is not arbitrable by it), the American Arbitration Association (the "AAA") in accordance with the commercial arbitration rules of the AAA. Prior to arbitration, all claims maintained by you must first be submitted to the Committee in accordance with claims procedures determined by the Committee. This paragraph is subject to the provisions of Paragraphs 15(b) and (c) below.
- (b) THE COMPANY AND YOU HEREBY IRREVOCABLY SUBMIT TO THE EXCLUSIVE JURISDICTION OF ANY STATE OR FEDERAL COURT LOCATED IN THE CITY OF NEW YORK OVER ANY SUIT, ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO OR CONCERNING THE PLAN OR THIS AWARD AGREEMENT THAT IS NOT OTHERWISE ARBITRATED OR RESOLVED ACCORDING TO PARAGRAPH 15(a) OF THIS AWARD AGREEMENT. This includes any suit, action or proceeding to compel arbitration or to enforce an arbitration award. The Company and you acknowledge that the forum designated by this Paragraph 15(b) has a reasonable relation to the Plan, this Award Agreement, and to your relationship

with the Company. Notwithstanding the foregoing, nothing herein shall preclude the Company from bringing any action, suit or proceeding in any other court for the purpose of enforcing the provisions of this Paragraph 15.

- (c) The agreement by you and the Company as to forum is independent of the law that may be applied in the action, suit or proceeding and you and the Company agree to such forum even if the forum may under applicable law choose to apply non-forum law. You and the Company hereby waive, to the fullest extent permitted by applicable law, any objection which you or the Company now or hereafter may have to personal jurisdiction or to the laying of venue of any such suit, action or proceeding in any court referred to in Paragraph 15(b). You and the Company undertake not to commence any action, suit or proceeding arising out of or relating to or concerning this Award Agreement in any forum other than a forum described in this Paragraph 15. You and (subject to the last sentence of Paragraph 15(b)) the Company agree that, to the fullest extent permitted by applicable law, a final and non-appealable judgment in any such suit, action or proceeding in any such court shall be conclusive and binding upon you and the Company.
- (d) You irrevocably appoint the Secretary of AIG as your agent for service of process in connection with any action, suit or proceeding arising out of or relating to or concerning this Award Agreement which is not arbitrated pursuant to the provisions of Paragraph 15(a), who shall promptly advise you of any such service of process.
- (e) You hereby agree to keep confidential the existence of, and any information concerning, a dispute described in this Paragraph 15, except that you may disclose information concerning such dispute to the arbitrator or court that is considering such dispute or to your legal counsel (provided that such counsel agrees not to disclose any such information other than as necessary to the prosecution or defense of the dispute).
- (f) You recognize and agree that prior to the grant of this Award you have no right to any benefits hereunder. Accordingly, in consideration of the receipt of this Award, you expressly waive any right to contest the amount of this Award, terms of this Award Agreement, any determination, action or omission hereunder or under the Plan by the Committee, AIG or the Board, or any amendment to the Plan or this Award Agreement (other than an amendment to which your consent is expressly required by Paragraph 13) and you expressly waive any claim related in any way to the Award including any claim based on any promissory estoppel or other theory in connection with this Award and your Employment with the Company.
- 16. Governing Law. THIS AWARD SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK, WITHOUT REGARD TO PRINCIPLES OF CONFLICTS OF LAWS.
- 17. <u>Headings</u>. The headings in this Award Agreement are for the purpose of convenience only and are not intended to define or limit the construction of the provisions hereof.

IN WITNESS WHER Date of Grant.	EOF, AMERIC	CAN INTERNA	ATIONAL GRO	OUP, INC. has caused this Award A	greement to be duly executed and delivered as of the
				AME	ERICAN INTERNATIONAL GROUP, INC.
				By:	
				By:	
Recipient: Number of RSUs: Date of Grant: Scheduled Vesting Date:					
Receipt Acknowledged:					
Address:					
		Street			
	City,	State	Zip Code		
	Social Sec	curity No./Loca	l I.D. No.		

-6-

#### **Glossary of Terms**

Solely for purposes of this award of RSUs, the following terms shall have the meanings set forth below. Capitalized terms not defined in this Glossary of Terms shall have the meanings as used or defined in the applicable Award Agreement or the Plan.

"Cause" means (i) your conviction, whether following trial or by plea of guilty or nolo contendere (or similar plea), in a criminal proceeding (A) on a misdemeanor charge involving fraud, false statements or misleading omissions, wrongful taking, embezzlement, bribery, forgery, counterfeiting or extortion, or (B) on a felony charge or (C) on an equivalent charge to those in clauses (A) and (B) in jurisdictions which do not use those designations; (ii) your engaging in any conduct which constitutes an employment disqualification under applicable law (including statutory disqualification as defined under the Exchange Act); (iii) your failure to perform your duties to the Company; (iv) your violation of any securities or commodities laws, any rules or regulations issued pursuant to such laws, or the rules and regulations of any securities or commodities exchange or association of which AIG or any of its subsidiaries or affiliates is a member; (v) your violation of any Company policy concerning hedging or confidential or proprietary information, or your material violation of any other Company policy as in effect from time to time; (vi) your engaging in any act or making any statement which impairs, impugns, denigrates, disparages or negatively reflects upon the name, reputation or business interests of the Company; or (vii) your engaging in any conduct detrimental to the Company. The determination as to whether "Cause" has occurred shall be made by the Committee in its sole discretion. The Committee shall also have the authority in its sole discretion to waive the consequences under the Plan or any Award Agreement of the existence or occurrence of any of the events, acts or omissions constituting "Cause."

"*Client*" means any client or prospective client of the Company to whom you provided services, or for whom you transacted business, or whose identity became known to you in connection with your relationship with or Employment by the Company.

"Competitive Enterprise" means a business enterprise that (i) engages in any activity, or (ii) owns or controls a significant interest in any entity that engages in any activity, that, in either case, competes anywhere with any activity in which the Company is engaged. The activities covered by the previous sentence include, without limitation, all insurance and re-insurance and re-insurance-related activities, asset management, financial product activities (including, without limitation, derivative activities) and financial services in the United States and abroad.

"*Disability*" means "permanent disability" as defined in the American International Group, Inc. Group Long — Term Insurance Policy as in effect on the Date of Grant.

"Retirement" means "normal retirement" as defined in the American International Group, Inc. Retirement Plan as in effect on the Date of Grant.

"Solicit" means any direct or indirect communication of any kind whatsoever, regardless of by whom initiated, inviting, advising, encouraging or requesting any person or entity, in any manner, to take or refrain from taking any action.

# AMERICAN INTERNATIONAL GROUP, INC. AMENDED AND RESTATED 2007 STOCK INCENTIVE PLAN RSU AWARD AGREEMENT

This award agreement (this "Award Agreement") sets forth the terms and conditions of an award (this "Award") of restricted stock units ("RSUs") granted to you under the American International Group, Inc. Amended and Restated 2007 Stock Incentive Plan (the "Plan").

- 1. The Plan. This Award is made pursuant to the Plan, the terms of which are incorporated in this Award Agreement. Capitalized terms used in this Award Agreement that are not defined in this Award Agreement, or in the attached Glossary of Terms, have the meanings as used or defined in the Plan.
- 2. <u>Award</u>. The number of RSUs subject to this Award is set forth at the end of this Award Agreement. Each RSU constitutes an unfunded and unsecured promise of AIG to deliver (or cause to be delivered) to you, subject to the terms of this Award Agreement, one share of Common Stock (the "*Share*" or the "*Shares*" as the context requires) (or securities or other property equal to the Fair Market Value thereof) on the Scheduled Vesting Date as provided herein. Until such delivery, you have only the rights of a general unsecured creditor, and no rights as a shareholder, of AIG. THIS AWARD IS SUBJECT TO ALL TERMS, CONDITIONS AND PROVISIONS OF THE PLAN AND THIS AWARD AGREEMENT, INCLUDING, WITHOUT LIMITATION, THE ARBITRATION AND CHOICE OF FORUM PROVISIONS SET FORTH IN PARAGRAPH 15.
  - 3. Vesting and Delivery.
- (a) <u>Vesting</u>. Except as provided in this Paragraph 3 and in Paragraphs 4 and 6, you shall become vested in the RSUs, and the Shares underlying the RSUs shall be delivered, on the fourth anniversary of the Date of Grant specified at the end of this Award Agreement (the "*Scheduled Vesting Date*"). Except as provided in Paragraph 6, if your Employment terminates for any reason prior to the Scheduled Vesting Date, your rights in respect of all of your RSUs shall terminate, and no Shares (or securities or other property) shall be delivered in respect of such RSUs.
- (b) <u>Delivery</u>. Except as provided in this Paragraph 3 and in Paragraphs 4, 6, 8 and 9, the Shares underlying the vested RSUs shall be delivered on the Scheduled Vesting Date. The Company may, at its option, deliver securities or other property in lieu of all or any portion of the Shares otherwise deliverable on the Scheduled Vesting Date. Such payment shall equal the product of the number of Shares to be delivered on the Scheduled Vesting Date and the Fair Market Value of one Share of Common Stock on the Scheduled Vesting Date. You shall be deemed the beneficial owner of the Shares at the close of business on the Scheduled Vesting Date and shall be entitled to any dividend or distribution that has not already been made with respect to such Shares if the record date for such dividend or distribution is on or after the close of business on the Scheduled Vesting Date. Notwithstanding the foregoing, if the Scheduled Vesting Date occurs at a time when you are considered by AIG to be one of its "covered employees" within the meaning of Section 162(m) of the Code, then, unless the Committee determines otherwise, delivery of the Shares (or securities or other property) automatically shall be deferred until after you have ceased to be such a covered employee.

(c) <u>Death</u>. Notwithstanding any other provision of this Award Agreement, if you die prior to the Scheduled Vesting Date, and provided your rights in respect of your RSUs have not previously terminated, the Shares (or securities or other property in lieu of all or any part thereof) corresponding to your outstanding RSUs shall be delivered to the representative of your estate as soon as practicable after the date of death and after such documentation as may be requested by the Committee is provided to the Committee.

### 4. Termination of RSUs and Non-Delivery of Shares.

- (a) Except as provided in Paragraphs 3(c) and 6, your rights in respect of your outstanding RSUs shall immediately terminate, and no Shares (or securities or other property) shall be delivered in respect of such unvested RSUs, if at any time prior to the Scheduled Vesting Date your Employment with the Company terminates for any reason, or you are otherwise no longer actively Employed by the Company.
- (b) Unless the Committee determines otherwise, and except as further provided in Paragraph 6, your rights in respect of all of your RSUs (whether or not vested) shall immediately terminate, and no Shares (or securities or other property) shall be delivered in respect of such RSUs, if at any time prior to the Scheduled Vesting Date:
  - (i) you attempt to have any dispute under this Award Agreement or the Plan resolved in any manner that is not provided for by Paragraph 15; or
  - (ii) any event that constitutes Cause has occurred; or
  - (iii) you in any manner, directly or indirectly, (A) Solicit any Client to transact business with a Competitive Enterprise or to reduce or refrain from doing any business with the Company or (B) interfere with or damage (or attempt to interfere with or damage) any relationship between the Company and any such Client or (C) Solicit any person who is an employee of the Company to resign from the Company or to apply for or accept employment with any Competitive Enterprise; or
  - (iv) you fail to certify to AIG, in accordance with procedures established by the Committee, with respect to the Scheduled Vesting Date that you have complied, or the Committee determines that you have failed as of the Scheduled Vesting Date to comply, with all of the terms and conditions of this Award Agreement. By accepting the delivery of Shares (or securities or other property) under this Award Agreement, you shall be deemed to have represented and certified at such time that you have complied with all the terms and conditions of this Award Agreement.
- (c) Unless the Committee determines otherwise, if the Scheduled Vesting Date in respect of any of your outstanding RSUs occurs, and Shares (or securities or other property) with respect to such outstanding RSUs would be deliverable under the terms and conditions of this Award Agreement, except that you have not complied with the conditions or your obligations under Paragraph 4(b)(iv), all of your rights with respect to your outstanding RSUs shall terminate no later than the Scheduled Vesting Date for such Shares.

5. <u>Repayment</u>. If, following the delivery of Shares (or securities or other property), the Committee determines that all terms and conditions of this Award Agreement in respect of such delivery were not satisfied, the Company shall be entitled to receive, and you shall be obligated to pay the Company immediately upon demand therefor, the Fair Market Value of the Shares (determined as of the Scheduled Vesting Date) and the amount of securities or other property (to the extent that any securities or other property was delivered in lieu of Shares) delivered with respect to the Scheduled Vesting Date, without reduction for any Shares (or securities or other property) applied to satisfy withholding tax or other obligations in respect of such Shares (or securities or other property).

#### 6. Disability and Retirement.

- (a) Notwithstanding any other provision of this Award Agreement, but subject to Paragraph 6(b):
- (i) if your Employment with the Company is terminated by reason of Disability or Retirement, the condition set forth in Paragraph 4(a) shall be waived with respect to your then outstanding unvested RSUs (as a result of which any such then unvested outstanding RSUs shall vest and shares corresponding to the RSUs shall be delivered to you as soon as practicable after the date of termination and after such documentation as may be requested by the Committee is provided to the Committee), but all other conditions of this Award Agreement shall continue to apply; and
- (ii) if your Employment with the Company is terminated by reason of Early Retirement, the condition set forth in Paragraph 4(a) shall be waived with respect to a fraction of your then outstanding unvested RSUs, the numerator of which fraction shall be the number of whole months between the Date of Grant and the date of such termination of Employment by reason of Early Retirement, and the denominator of which fraction shall be the number of whole months between the Date of Grant and the Scheduled Vesting Date (as a result of which such fraction of any such then unvested outstanding RSUs shall vest, and shares corresponding to such fraction of the RSUs shall be delivered to you as soon as practicable after the date of termination and after such documentation as may be requested by the Committee is provided to the Committee, but all other unvested outstanding RSUs shall immediately terminate), but all other conditions of this Award Agreement shall continue to apply.
- (b) Without limiting the application of Paragraph 4(b) or Paragraph 4(c), your rights in respect of any outstanding RSUs that become vested solely by reason of Paragraph 6(a) shall terminate, and no Shares (or securities or other property) shall be delivered in respect of such outstanding RSUs if, following the termination of your Employment with the Company by reason of Disability, Retirement or Early Retirement and prior to the Scheduled Vesting Date, you (i) form, or acquire a 5% or greater equity ownership, voting or profit participation interest in, any Competitive Enterprise or (ii) associate in any capacity (including, but not limited to, association as an officer, employee, partner, director, consultant, agent or advisor) with any Competitive Enterprise.
- 7. Non-transferability. Except as otherwise may be provided by the Committee, the limitations set forth in Section 3.5 of the Plan shall apply. Any assignment or delegation in violation of the provisions of this Paragraph 7 shall be null and void.

- 8. Withholding, Consents and Legends.
- (a) The delivery of Shares is conditioned on your satisfaction of any applicable withholding taxes (in accordance with Section 3.2 of the Plan).
- (b) Your rights in respect of your RSUs are conditioned on the receipt to the full satisfaction of the Committee of any required Consent (as defined in Section 3.3 of the Plan) that the Committee may determine to be necessary or advisable (including, without limitation, your consenting to deductions from your wages, or another arrangement satisfactory to the Committee, to reimburse the Company for advances made on your behalf to satisfy withholding and other tax obligations in connection with this Award).
- (c) AIG may affix to Certificates representing Shares issued pursuant to this Award Agreement any legend that the Committee determines to be necessary or advisable (including to reflect any restrictions to which you may be subject under a separate agreement with AIG). AIG may advise the transfer agent to place a stop transfer order against any legended Shares.
- 9. <u>Right of Offset</u>. The Company shall have the right to offset against the obligation to deliver Shares (or securities or other property) under this Award Agreement any outstanding amounts (including, without limitation, travel and entertainment or advance account balances, loans, or amounts repayable to the Company pursuant to tax equalization, housing, automobile or other employee programs) you then owe to the Company and any amounts the Committee otherwise deems appropriate.
- 10. No Rights to Continued Employment. Nothing in this Award Agreement or the Plan shall be construed as giving you any right to continued Employment by the Company or affect any right that the Company may have to terminate or alter the terms and conditions of your Employment.
- 11. <u>Successors and Assigns of AIG</u>. The terms and conditions of this Award Agreement shall be binding upon, and shall inure to the benefit of, AIG and its successor entities (as defined in Section 3.6 of the Plan).
- 12. <u>Committee Discretion</u>. The Committee shall have full discretion with respect to any actions to be taken or determinations to be made in connection with this Award Agreement, and its determinations shall be final, binding and conclusive.
- 13. <u>Amendment</u>. The Committee reserves the right at any time to amend the terms and conditions set forth in this Award Agreement, and the Board may amend the Plan in any respect; provided, that, notwithstanding the foregoing and Sections 1.3.1(a), 1.3.1(b) and 3.1 of the Plan, no such amendment shall materially adversely affect your rights and obligations under this Award Agreement without your consent. Any amendment of this Award Agreement shall be in writing signed by an authorized member of the Committee or a person or persons designated by the Committee.
  - 14. Adjustment. The RSUs shall be subject to amendment and adjustment in accordance with Section 1.6.4 of the Plan.
  - 15. Arbitration; Choice of Forum.

- (a) Any dispute, controversy or claim between the Company and you, arising out of or relating to or concerning the Plan or this Award Agreement, shall be finally settled by arbitration in New York City before, and in accordance with the rules then obtaining of, the New York Stock Exchange, Inc. (the "NYSE") or, if the NYSE declines to arbitrate the matter (or if the matter otherwise is not arbitrable by it), the American Arbitration Association (the "AAA") in accordance with the commercial arbitration rules of the AAA. Prior to arbitration, all claims maintained by you must first be submitted to the Committee in accordance with claims procedures determined by the Committee. This paragraph is subject to the provisions of Paragraphs 15(b) and (c) below.
- (b) THE COMPANY AND YOU HEREBY IRREVOCABLY SUBMIT TO THE EXCLUSIVE JURISDICTION OF ANY STATE OR FEDERAL COURT LOCATED IN THE CITY OF NEW YORK OVER ANY SUIT, ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO OR CONCERNING THE PLAN OR THIS AWARD AGREEMENT THAT IS NOT OTHERWISE ARBITRATED OR RESOLVED ACCORDING TO PARAGRAPH 15(a) OF THIS AWARD AGREEMENT. This includes any suit, action or proceeding to compel arbitration or to enforce an arbitration award. The Company and you acknowledge that the forum designated by this Paragraph 15(b) has a reasonable relation to the Plan, this Award Agreement, and to your relationship with the Company. Notwithstanding the foregoing, nothing herein shall preclude the Company from bringing any action, suit or proceeding in any other court for the purpose of enforcing the provisions of this Paragraph 15.
- (c) The agreement by you and the Company as to forum is independent of the law that may be applied in the action, suit or proceeding and you and the Company agree to such forum even if the forum may under applicable law choose to apply non-forum law. You and the Company hereby waive, to the fullest extent permitted by applicable law, any objection which you or the Company now or hereafter may have to personal jurisdiction or to the laying of venue of any such suit, action or proceeding in any court referred to in Paragraph 15(b). You and the Company undertake not to commence any action, suit or proceeding arising out of or relating to or concerning this Award Agreement in any forum other than a forum described in this Paragraph 15. You and (subject to the last sentence of Paragraph 15(b)) the Company agree that, to the fullest extent permitted by applicable law, a final and non-appealable judgment in any such suit, action or proceeding in any such court shall be conclusive and binding upon you and the Company.
- (d) You irrevocably appoint the Secretary of AIG as your agent for service of process in connection with any action, suit or proceeding arising out of or relating to or concerning this Award Agreement which is not arbitrated pursuant to the provisions of Paragraph 15(a), who shall promptly advise you of any such service of process.
- (e) You hereby agree to keep confidential the existence of, and any information concerning, a dispute described in this Paragraph 15, except that you may disclose information concerning such dispute to the arbitrator or court that is considering such dispute or to your legal counsel (provided that such counsel agrees not to disclose any such information other than as necessary to the prosecution or defense of the dispute).
- (f) You recognize and agree that prior to the grant of this Award you have no right to any benefits hereunder. Accordingly, in consideration of the receipt of this Award, you expressly waive any right to contest the amount of this Award, terms of this Award Agreement, any determination, action or omission hereunder or under the

Plan by the Committee, AIG or the Board, or any amendment to the Plan or this Award Agreement (other than an amendment to which your consent is expressly required by Paragraph 13) and you expressly waive any claim related in any way to the Award including any claim based on any promissory estoppel or other theory in connection with this Award and your Employment with the Company.

- 16. <u>Governing Law</u>. THIS AWARD SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK, WITHOUT REGARD TO PRINCIPLES OF CONFLICTS OF LAWS.
- 17. <u>Headings</u>. The headings in this Award Agreement are for the purpose of convenience only and are not intended to define or limit the construction of the provisions hereof.

IN WITNESS WHER	EOF, AMERI	CAN INTERN	ATIONAL GRO	OUP, INC. has caused this Award A	Agreement to be duly executed and delivered as of the
Date of Grant.					
				AM	IERICAN INTERNATIONAL GROUP, INC.
				By:	
				Ву:	
Recipient: Number of RSUs:					
Date of Grant:					
Scheduled Vesting Date:					
Receipt Acknowledged:					
Address:					
		Street			
	City,	State	Zip Code		
	Social Se	curity No./Loc	al I.D. No.		

-7-

#### **Glossary of Terms**

Solely for purposes of this award of RSUs, the following terms shall have the meanings set forth below. Capitalized terms not defined in this Glossary of Terms shall have the meanings as used or defined in the applicable Award Agreement or the Plan.

"Cause" means (i) your conviction, whether following trial or by plea of guilty or nolo contendere (or similar plea), in a criminal proceeding (A) on a misdemeanor charge involving fraud, false statements or misleading omissions, wrongful taking, embezzlement, bribery, forgery, counterfeiting or extortion, or (B) on a felony charge or (C) on an equivalent charge to those in clauses (A) and (B) in jurisdictions which do not use those designations; (ii) your engaging in any conduct which constitutes an employment disqualification under applicable law (including statutory disqualification as defined under the Exchange Act); (iii) your failure to perform your duties to the Company; (iv) your violation of any securities or commodities laws, any rules or regulations issued pursuant to such laws, or the rules and regulations of any securities or commodities exchange or association of which AIG or any of its subsidiaries or affiliates is a member; (v) your violation of any Company policy concerning hedging or confidential or proprietary information, or your material violation of any other Company policy as in effect from time to time; (vi) your engaging in any act or making any statement which impairs, impugns, denigrates, disparages or negatively reflects upon the name, reputation or business interests of the Company; or (vii) your engaging in any conduct detrimental to the Company. The determination as to whether "Cause" has occurred shall be made by the Committee in its sole discretion. The Committee shall also have the authority in its sole discretion to waive the consequences under the Plan or any Award Agreement of the existence or occurrence of any of the events, acts or omissions constituting "Cause."

"Client" means any client or prospective client of the Company to whom you provided services, or for whom you transacted business, or whose identity became known to you in connection with your relationship with or Employment by the Company.

"Competitive Enterprise" means a business enterprise that (i) engages in any activity, or (ii) owns or controls a significant interest in any entity that engages in any activity, that, in either case, competes anywhere with any activity in which the Company is engaged. The activities covered by the previous sentence include, without limitation, all insurance and re-insurance and re-insurance-related activities, asset management, financial product activities (including, without limitation, derivative activities) and financial services in the United States and abroad.

"Disability" means "permanent disability" as defined in the American International Group, Inc. Group Long — Term Insurance Policy as in effect on the Date of Grant.

"*Early Retirement*" means retirement from Employment with the Company before Retirement but after reaching age 55, with the approval of, and subject to any terms and conditions prescribed by, the Committee in its sole discretion.

"Retirement" means "normal retirement" as defined in the American International Group, Inc. Retirement Plan as in effect on the Date of Grant.

"Solicit" means any direct or indirect communication of any kind whatsoever, regardless of by whom initiated, inviting, advising, encouraging or requesting any person or entity, in any manner, to take or refrain from taking any action.

# AMERICAN INTERNATIONAL GROUP, INC. AMENDED AND RESTATED 2007 STOCK INCENTIVE PLAN RSU AWARD AGREEMENT

This award agreement (this "Award Agreement") sets forth the terms and conditions of an award (this "Award") of restricted stock units ("RSUs") granted to you under the American International Group, Inc. Amended and Restated 2007 Stock Incentive Plan (the "Plan").

- 1. The Plan. This Award is made pursuant to the Plan, the terms of which are incorporated in this Award Agreement. Capitalized terms used in this Award Agreement that are not defined in this Award Agreement, or in the attached Glossary of Terms, have the meanings as used or defined in the Plan.
- 2. <u>Award</u>. The number of RSUs subject to this Award is set forth at the end of this Award Agreement. Each RSU constitutes an unfunded and unsecured promise of AIG to deliver (or cause to be delivered) to you, subject to the terms of this Award Agreement, one share of Common Stock (the "Share" or the "Shares" as the context requires) (or securities or other property equal to the Fair Market Value thereof) on the Scheduled Vesting Date as provided herein. Until such delivery, you have only the rights of a general unsecured creditor, and no rights as a shareholder, of AIG. THIS AWARD IS SUBJECT TO ALL TERMS, CONDITIONS AND PROVISIONS OF THE PLAN AND THIS AWARD AGREEMENT, INCLUDING, WITHOUT LIMITATION, THE ARBITRATION AND CHOICE OF FORUM PROVISIONS SET FORTH IN PARAGRAPH 15.
  - 3. Vesting and Delivery.
- (a) <u>Vesting</u>. Except as provided in this Paragraph 3 and in Paragraphs 4 and 6, you shall become vested in the RSUs, and the Shares underlying the RSUs shall be delivered, on the third anniversary of the Date of Grant specified at the end of this Award Agreement (the "*Scheduled Vesting Date*"). Except as provided in Paragraph 6, if your Employment terminates for any reason prior to the Scheduled Vesting Date, your rights in respect of all of your RSUs shall terminate, and no Shares (or securities or other property) shall be delivered in respect of such RSUs.
- (b) <u>Delivery</u>. Except as provided in this Paragraph 3 and in Paragraphs 4, 6, 8 and 9, the Shares underlying the vested RSUs shall be delivered on the Scheduled Vesting Date. The Company may, at its option, deliver securities or other property in lieu of all or any portion of the Shares otherwise deliverable on the Scheduled Vesting Date. Such payment shall equal the product of the number of Shares to be delivered on the Scheduled Vesting Date and the Fair Market Value of one Share of Common Stock on the Scheduled Vesting Date. You shall be deemed the beneficial owner of the Shares at the close of business on the Scheduled Vesting Date and shall be entitled to any dividend or distribution that has not already been made with respect to such Shares if the record date for such dividend or distribution is on or after the close of business on the Scheduled Vesting Date. Notwithstanding the foregoing, if the Scheduled Vesting Date occurs at a time when you are considered by AIG to be one of its "covered employees" within the meaning of Section 162(m) of the Code, then, unless the Committee determines otherwise, delivery of the Shares (or securities or other

property) automatically shall be deferred until after you have ceased to be such a covered employee.

(c) <u>Death</u>. Notwithstanding any other provision of this Award Agreement, if you die prior to the Scheduled Vesting Date, and provided your rights in respect of your RSUs have not previously terminated, the Shares (or securities or other property in lieu of all or any part thereof) corresponding to your outstanding RSUs shall be delivered to the representative of your estate as soon as practicable after the date of death and after such documentation as may be requested by the Committee is provided to the Committee.

### 4. Termination of RSUs and Non-Delivery of Shares.

- (a) Except as provided in Paragraphs 3(c) and 6, your rights in respect of your outstanding RSUs shall immediately terminate, and no Shares (or securities or other property) shall be delivered in respect of such unvested RSUs, if at any time prior to the Scheduled Vesting Date your Employment with the Company terminates for any reason, or you are otherwise no longer actively Employed by the Company.
- (b) Unless the Committee determines otherwise, and except as further provided in Paragraph 6, your rights in respect of all of your RSUs (whether or not vested) shall immediately terminate, and no Shares (or securities or other property) shall be delivered in respect of such RSUs, if at any time prior to the Scheduled Vesting Date:
  - (i) you attempt to have any dispute under this Award Agreement or the Plan resolved in any manner that is not provided for by Paragraph 15; or
  - (ii) any event that constitutes Cause has occurred; or
  - (iii) you in any manner, directly or indirectly, (A) Solicit any Client to transact business with a Competitive Enterprise or to reduce or refrain from doing any business with the Company or (B) interfere with or damage (or attempt to interfere with or damage) any relationship between the Company and any such Client or (C) Solicit any person who is an employee of the Company to resign from the Company or to apply for or accept employment with any Competitive Enterprise; or
  - (iv) you fail to certify to AIG, in accordance with procedures established by the Committee, with respect to the Scheduled Vesting Date that you have complied, or the Committee determines that you have failed as of the Scheduled Vesting Date to comply, with all of the terms and conditions of this Award Agreement. By accepting the delivery of Shares (or securities or other property) under this Award Agreement, you shall be deemed to have represented and certified at such time that you have complied with all the terms and conditions of this Award Agreement.
- (c) Unless the Committee determines otherwise, if the Scheduled Vesting Date in respect of any of your outstanding RSUs occurs, and Shares (or securities or other property) with respect to such outstanding RSUs would be deliverable under the terms and conditions of this Award Agreement, except that you have not complied with the conditions or your obligations under Paragraph 4(b)(iv), all

of your rights with respect to your outstanding RSUs shall terminate no later than the Scheduled Vesting Date for such Shares.

5. Repayment. If, following the delivery of Shares (or securities or other property), the Committee determines that all terms and conditions of this Award Agreement in respect of such delivery were not satisfied, the Company shall be entitled to receive, and you shall be obligated to pay the Company immediately upon demand therefor, the Fair Market Value of the Shares (determined as of the Scheduled Vesting Date) and the amount of securities or other property (to the extent that any securities or other property was delivered in lieu of Shares) delivered with respect to the Scheduled Vesting Date, without reduction for any Shares (or securities or other property) applied to satisfy withholding tax or other obligations in respect of such Shares (or securities or other property).

#### 6. Disability and Retirement.

- (a) Notwithstanding any other provision of this Award Agreement, but subject to Paragraph 6(b):
- (i) if your Employment with the Company is terminated by reason of Disability or Retirement following the first anniversary of the Date of Grant, the condition set forth in Paragraph 4(a) shall be waived with respect to your then outstanding unvested RSUs (as a result of which any such then unvested outstanding RSUs shall vest and shares corresponding to the RSUs shall be delivered to you as soon as practicable after the date of termination and after such documentation as may be requested by the Committee is provided to the Committee), but all other conditions of this Award Agreement shall continue to apply; and
- (ii) if your Employment with the Company is terminated by reason of Retirement prior to December 31. 20\_\_\_\_, the condition set forth in Paragraph 4(a) shall be waived only with respect to a fraction of your then outstanding unvested RSUs, the numerator of which fraction shall be the number of whole quarters between January 1, 20\_\_\_and the date of such termination of Employment by reason of Retirement, and the denominator of which fraction shall be four (as a result of which such fraction of any such then unvested outstanding RSUs shall vest, and shares corresponding to such fraction of the RSUs shall be delivered to you as soon as practicable after the date of termination and after such documentation as may be requested by the Committee is provided to the Committee, but all other unvested outstanding RSUs shall immediately terminate), but all other conditions of this Award Agreement shall continue to apply.
- (iii) if your Employment with the Company is terminated by reason of Early Retirement, the condition set forth in Paragraph 4(a) shall be waived with respect to a fraction of your then outstanding unvested RSUs, the numerator of which fraction shall be the number of whole quarters between January 1, 20\_\_\_\_and the date of such termination of Employment by reason of Early Retirement, and the denominator of which fraction shall be twelve (as a result of which such fraction of any such then unvested outstanding RSUs shall vest, and shares corresponding to such fraction of the RSUs shall be delivered to you as soon as practicable after the date of termination and after such documentation as may be requested by the Committee is provided to the Committee, but all other unvested outstanding RSUs shall immediately

terminate), but all other conditions of this Award Agreement shall continue to apply.

- (b) Without limiting the application of Paragraph 4(b) or Paragraph 4(c), your rights in respect of any outstanding RSUs that become vested solely by reason of Paragraph 6(a) shall terminate, and no Shares (or securities or other property) shall be delivered in respect of such outstanding RSUs if, following the termination of your Employment with the Company by reason of Disability, Retirement or Early Retirement and prior to the Scheduled Vesting Date, you (i) form, or acquire a 5% or greater equity ownership, voting or profit participation interest in, any Competitive Enterprise or (ii) associate in any capacity (including, but not limited to, association as an officer, employee, partner, director, consultant, agent or advisor) with any Competitive Enterprise.
- 7. <u>Non-transferability</u>. Except as otherwise may be provided by the Committee, the limitations set forth in Section 3.5 of the Plan shall apply. Any assignment or delegation in violation of the provisions of this Paragraph 7 shall be null and void.
  - 8. Withholding, Consents and Legends.
  - (a) The delivery of Shares is conditioned on your satisfaction of any applicable withholding taxes (in accordance with Section 3.2 of the Plan).
- (b) Your rights in respect of your RSUs are conditioned on the receipt to the full satisfaction of the Committee of any required Consent (as defined in Section 3.3 of the Plan) that the Committee may determine to be necessary or advisable (including, without limitation, your consenting to deductions from your wages, or another arrangement satisfactory to the Committee, to reimburse the Company for advances made on your behalf to satisfy withholding and other tax obligations in connection with this Award).
- (c) AIG may affix to Certificates representing Shares issued pursuant to this Award Agreement any legend that the Committee determines to be necessary or advisable (including to reflect any restrictions to which you may be subject under a separate agreement with AIG). AIG may advise the transfer agent to place a stop transfer order against any legended Shares.
- 9. <u>Right of Offset</u>. The Company shall have the right to offset against the obligation to deliver Shares (or securities or other property) under this Award Agreement any outstanding amounts (including, without limitation, travel and entertainment or advance account balances, loans, or amounts repayable to the Company pursuant to tax equalization, housing, automobile or other employee programs) you then owe to the Company and any amounts the Committee otherwise deems appropriate.
- 10. No Rights to Continued Employment. Nothing in this Award Agreement or the Plan shall be construed as giving you any right to continued Employment by the Company or affect any right that the Company may have to terminate or alter the terms and conditions of your Employment.
- 11. <u>Successors and Assigns of AIG</u>. The terms and conditions of this Award Agreement shall be binding upon, and shall inure to the benefit of, AIG and its successor entities (as defined in Section 3.6 of the Plan).

- 12. <u>Committee Discretion</u>. The Committee shall have full discretion with respect to any actions to be taken or determinations to be made in connection with this Award Agreement, and its determinations shall be final, binding and conclusive.
- 13. <u>Amendment</u>. The Committee reserves the right at any time to amend the terms and conditions set forth in this Award Agreement, and the Board may amend the Plan in any respect; provided, that, notwithstanding the foregoing and Sections 1.3.1(a), 1.3.1(b) and 3.1 of the Plan, no such amendment shall materially adversely affect your rights and obligations under this Award Agreement without your consent. Any amendment of this Award Agreement shall be in writing signed by an authorized member of the Committee or a person or persons designated by the Committee.
  - 14. Adjustment. The RSUs shall be subject to amendment and adjustment in accordance with Section 1.6.4 of the Plan.
  - 15. Arbitration; Choice of Forum.
- (a) Any dispute, controversy or claim between the Company and you, arising out of or relating to or concerning the Plan or this Award Agreement, shall be finally settled by arbitration in New York City before, and in accordance with the rules then obtaining of, the New York Stock Exchange, Inc. (the "NYSE") or, if the NYSE declines to arbitrate the matter (or if the matter otherwise is not arbitrable by it), the American Arbitration Association (the "AAA") in accordance with the commercial arbitration rules of the AAA. Prior to arbitration, all claims maintained by you must first be submitted to the Committee in accordance with claims procedures determined by the Committee. This paragraph is subject to the provisions of Paragraphs 15(b) and (c) below.
- (b) THE COMPANY AND YOU HEREBY IRREVOCABLY SUBMIT TO THE EXCLUSIVE JURISDICTION OF ANY STATE OR FEDERAL COURT LOCATED IN THE CITY OF NEW YORK OVER ANY SUIT, ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO OR CONCERNING THE PLAN OR THIS AWARD AGREEMENT THAT IS NOT OTHERWISE ARBITRATED OR RESOLVED ACCORDING TO PARAGRAPH 15(a) OF THIS AWARD AGREEMENT. This includes any suit, action or proceeding to compel arbitration or to enforce an arbitration award. The Company and you acknowledge that the forum designated by this Paragraph 15(b) has a reasonable relation to the Plan, this Award Agreement, and to your relationship with the Company. Notwithstanding the foregoing, nothing herein shall preclude the Company from bringing any action, suit or proceeding in any other court for the purpose of enforcing the provisions of this Paragraph 15.
- (c) The agreement by you and the Company as to forum is independent of the law that may be applied in the action, suit or proceeding and you and the Company agree to such forum even if the forum may under applicable law choose to apply non-forum law. You and the Company hereby waive, to the fullest extent permitted by applicable law, any objection which you or the Company now or hereafter may have to personal jurisdiction or to the laying of venue of any such suit, action or proceeding in any court referred to in Paragraph 15(b). You and the Company undertake not to commence any action, suit or proceeding arising out of or relating to or concerning this Award Agreement in any forum other than a forum described in this Paragraph 15. You and (subject to the last sentence of Paragraph 15(b)) the Company agree that, to the fullest extent permitted by applicable law, a final and non-appealable judgment in any such suit, action or proceeding in any such court shall be conclusive and binding upon you and the Company.

- (d) You irrevocably appoint the Secretary of AIG as your agent for service of process in connection with any action, suit or proceeding arising out of or relating to or concerning this Award Agreement which is not arbitrated pursuant to the provisions of Paragraph 15(a), who shall promptly advise you of any such service of process.
- (e) You hereby agree to keep confidential the existence of, and any information concerning, a dispute described in this Paragraph 15, except that you may disclose information concerning such dispute to the arbitrator or court that is considering such dispute or to your legal counsel (provided that such counsel agrees not to disclose any such information other than as necessary to the prosecution or defense of the dispute).
- (f) You recognize and agree that prior to the grant of this Award you have no right to any benefits hereunder. Accordingly, in consideration of the receipt of this Award, you expressly waive any right to contest the amount of this Award, terms of this Award Agreement, any determination, action or omission hereunder or under the Plan by the Committee, AIG or the Board, or any amendment to the Plan or this Award Agreement (other than an amendment to which your consent is expressly required by Paragraph 13) and you expressly waive any claim related in any way to the Award including any claim based on any promissory estoppel or other theory in connection with this Award and your Employment with the Company.
- 16. <u>Governing Law</u>. THIS AWARD SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK, WITHOUT REGARD TO PRINCIPLES OF CONFLICTS OF LAWS.
- 17. <u>Headings</u>. The headings in this Award Agreement are for the purpose of convenience only and are not intended to define or limit the construction of the provisions hereof.

IN WITNESS WHEREO Date of Grant.	DF, AMERICAN IN	TERNATIONAL GR	OUP, INC. has caused th	nis Award Agreement to be duly	executed and delivered as of the
			AMERICAN INTE	RNATIONAL GROUP, INC.	
			Ву:		
			Ву:		
Recipient:					
Number of RSUs:					
Date of Grant:					
Scheduled Vesting Date:					
Receipt Acknowledged:					
Address:					
		Street			
	City,	State	Zip Code		
	Social Security I	No./Local I.D. No.			
			-7-		

#### **Glossary of Terms**

Solely for purposes of this award of RSUs, the following terms shall have the meanings set forth below. Capitalized terms not defined in this Glossary of Terms shall have the meanings as used or defined in the applicable Award Agreement or the Plan.

"Cause" means (i) your conviction, whether following trial or by plea of guilty or nolo contendere (or similar plea), in a criminal proceeding (A) on a misdemeanor charge involving fraud, false statements or misleading omissions, wrongful taking, embezzlement, bribery, forgery, counterfeiting or extortion, or (B) on a felony charge or (C) on an equivalent charge to those in clauses (A) and (B) in jurisdictions which do not use those designations; (ii) your engaging in any conduct which constitutes an employment disqualification under applicable law (including statutory disqualification as defined under the Exchange Act); (iii) your failure to perform your duties to the Company; (iv) your violation of any securities or commodities laws, any rules or regulations issued pursuant to such laws, or the rules and regulations of any securities or commodities exchange or association of which AIG or any of its subsidiaries or affiliates is a member; (v) your violation of any Company policy concerning hedging or confidential or proprietary information, or your material violation of any other Company policy as in effect from time to time; (vi) your engaging in any act or making any statement which impairs, impugns, denigrates, disparages or negatively reflects upon the name, reputation or business interests of the Company; or (vii) your engaging in any conduct detrimental to the Company. The determination as to whether "Cause" has occurred shall be made by the Committee in its sole discretion. The Committee shall also have the authority in its sole discretion to waive the consequences under the Plan or any Award Agreement of the existence or occurrence of any of the events, acts or omissions constituting "Cause."

"Client" means any client or prospective client of the Company to whom you provided services, or for whom you transacted business, or whose identity became known to you in connection with your relationship with or Employment by the Company.

"Competitive Enterprise" means a business enterprise that (i) engages in any activity, or (ii) owns or controls a significant interest in any entity that engages in any activity, that, in either case, competes anywhere with any activity in which the Company is engaged. The activities covered by the previous sentence include, without limitation, all insurance and re-insurance and re-insurance-related activities, asset management, financial product activities (including, without limitation, derivative activities) and financial services in the United States and abroad.

"Disability" means "permanent disability" as defined in the American International Group, Inc. Group Long — Term Insurance Policy as in effect on the Date of Grant.

"*Early Retirement*" means retirement from Employment with the Company before Retirement but after reaching age 55, with the approval of, and subject to any terms and conditions prescribed by, the Committee in its sole discretion.

"Retirement" means "normal retirement" as defined in the American International Group, Inc. Retirement Plan as in effect on the Date of Grant.

"Solicit" means any direct or indirect communication of any kind whatsoever, regardless of by whom initiated, inviting, advising, encouraging or requesting any person or entity, in any manner, to take or refrain from taking any action.

### AMERICAN INTERNATIONAL GROUP, INC. AMENDED AND RESTATED 2007 STOCK INCENTIVE PLAN NON-EMPLOYEE DIRECTOR DEFERRED STOCK UNITS AWARD AGREEMENT

This award agreement (this "Award Agreement") sets forth the terms and conditions of an award (this "Award") of deferred stock units ("DSUs") granted to you under the American International Group, Inc. Amended and Restated 2007 Stock Incentive Plan (as amended and restated from time to time, the "Plan").

- 1. <u>The Plan</u>. This Award is made pursuant to the Plan, the terms of which are incorporated in this Award Agreement. Capitalized terms used in this Award Agreement which are not defined in this Award Agreement have the meanings as used or defined in the Plan.
- 2. <u>Award</u>. The number of DSUs set forth on the signature page of this Award Agreement are subject to this Award. Each DSU constitutes an unfunded and unsecured promise of AIG to deliver (or cause to be delivered) to you, subject to the terms of this Award Agreement, one share of Common Stock (a "Share" or "Shares" as the context requires) (or securities or other property equal to the Fair Market Value thereof) on the Delivery Date as provided herein. Until such delivery, you have only the rights of a general unsecured creditor and no rights as a shareholder of AIG. In the event the calculation of the number of DSUs subject to this Award Agreement results in fractional shares, the number of Shares shall be rounded down to the next whole Share. **THIS AWARD IS SUBJECT TO ALL TERMS, CONDITIONS AND PROVISIONS OF THE PLAN AND THIS AWARD AGREEMENT.**

#### 3. Delivery.

- (a) In General. Except as provided below in this Paragraph 3 and Paragraph 8, the Delivery Date shall be on the last Trading Day of the month in which you cease to be a director of the Board. AIG may deliver securities or other property based on the Fair Market Value in lieu of all or any portion of the Shares otherwise deliverable on the Delivery Date. You shall be the beneficial owner of any Shares at the close of business on the Delivery Date and shall be entitled to any dividend or distribution that has not already been made with respect to such Shares if the record date for such dividend or distribution is on or after the close of business on the Delivery Date. "Trading Day" means a day on which the Common Stock trades regular way on the New York Stock Exchange or, if the Common Stock is not then listed on the New York Stock Exchange, on the principal national or regional securities exchange on which the Common Stock is then listed or, if the Common Stock is not then listed on a national or regional securities exchange, on the principal other market on which the Common Stock is then traded.
- (b) <u>Death</u>. Notwithstanding any other term or provision of this Award Agreement, if you die prior to the Delivery Date, the Shares (or securities or other property in lieu of all or any portion thereof) corresponding to your outstanding DSUs shall be delivered to the representative of your estate as soon as practicable after the date of death and after such documentation as may be requested by the Board is provided to the Board.

- 4. <u>Dividend Equivalent Rights</u>. On the first day of each quarter before the delivery of Shares (or securities or other property in lieu thereof) pursuant to this Award Agreement, you shall be entitled to receive, in the form of DSUs, an amount equal to any regular cash dividend payment that would have been distributed during the preceding quarter in respect of the Shares not yet delivered. The number of DSUs awarded under this Section 4 shall be based upon the closing price of a Share on the first day of the quarter succeeding the dividend payment date. In the event the calculation of the number of DSUs to be received results in fractional shares (after taking into account all DSUs to be awarded to you in respect of such date), the number of Shares shall be rounded down to the next whole Share. In the event that a regular cash dividend payment is paid during the quarter in which you cease to be a director (but prior to the date on which you cease to be a director) you will be entitled to receive an amount in cash on the Delivery Date equal to the actual dividend paid for that quarter in respect of the Shares.
- 5. Non-transferability. Except as may otherwise be provided in this Paragraph or as otherwise may be provided by the Board, the limitations set forth in Section 3.5 of the Plan shall apply. Any assignment in violation of the provisions of this Paragraph 5 shall be void. The Board may adopt procedures pursuant to which you may transfer some or all of your DSUs through a gift for no consideration to any child, stepchild, grandchild, parent, stepparent, grandparent, spouse, sibling, niece, nephew, mother-in-law, father-in-law, son-in-law, daughter-in-law, brother-in-law or sister-in-law, including adoptive relationships, any person sharing the recipient's household (other than a tenant or employee), a trust in which these persons have more than 50% of the beneficial interest, and any other entity in which these persons (or the recipient) own more than 50% of the voting interests.

#### 6. Withholding, Consents and Legends.

- (a) The delivery of Shares is conditioned on your satisfaction of any applicable withholding taxes in accordance with Section 3.2 of the Plan.
- (b) Your rights in respect of the DSUs are conditioned on the receipt to the full satisfaction of the Board of any required consents (as defined in Section 3.3 of the Plan) that the Board may determine to be necessary or advisable.
- (c) AIG may affix to Certificates representing Shares issued pursuant to this Award Agreement any legend that the Board determines to be necessary or advisable. AIG may advise the transfer agent to place a stop order against any legended Shares.
- 7. <u>Successors and Assigns of AIG</u>. The terms and conditions of this Award Agreement shall be binding upon and shall inure to the benefit of AIG and its successors and assigns.
- 8. <u>Amendment</u>. The Board reserves the right at any time to amend the terms and conditions set forth in this Award Agreement in any respect in accordance with Section 1.3 of the Plan, and the Board may amend the Plan in any respect in accordance with Section 3.1 of the Plan. Notwithstanding the foregoing and Sections 1.3.1 and 3.1 of the Plan, no such amendment shall materially adversely affect your rights and obligations under this Award Agreement without

your consent (or the consent of your estate, if such consent is obtained after your death), and the Board may not accelerate the delivery of the Shares (or securities or other property in lieu of any or all of the Shares). Any amendment of this Award Agreement shall be in writing signed by an authorized member of the Board or any other person or persons authorized by the Board.

- 9. <u>Governing Law</u>. THIS AWARD SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK, WITHOUT REGARD TO PRINCIPLES OF CONFLICT OF LAWS.
- 10. <u>Headings</u>. The headings in this Award Agreement are for the purpose of convenience only and are not intended to define or limit the construction of the provisions hereof.

IN WITNESS WHEREOF, AIG and you have caused this Award Agreement to be duly executed and delivered.

### AMERICAN INTERNATIONAL GROUP, INC.

	By: Name: Title:	
Recipient:	 -	
Number of DSUs:	 <del>-</del>	
Date:	 _	
Accepted and Agreed:		
By:		

#### CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

We hereby consent to the incorporation by reference in this Registration Statement on Form S-8 of our report dated March 1, 2007 relating to the consolidated financial statements, the financial statement schedules, management's assessment of the effectiveness of internal control over financial reporting and the effectiveness of internal control over financial reporting, which appears in American International Group, Inc.'s Annual Report on Form 10-K for the year ended December 31, 2006. We also consent to the reference to us under "Experts" included in the Prospectus, which forms a part of this Registration Statement on Form S-8.

PricewaterhouseCoopers LLP New York, New York December 19, 2007